DIET MANAGER By LEE CHEE WAH

A REPORT

SUBMITTED TO

Universiti Tunku Abdul Rahman

in partial fulfillment of the requirements

for the degree of

BACHELOR OF INFORMATION SYSTEMS (HONOURS) INFORMATION SYSTEMS

ENGINEERING

Faculty of Information and Communication Technology

(Kampar Campus)

JAN 2023

DIET MANAGER By LEE CHEE WAH

A REPORT

SUBMITTED TO

Universiti Tunku Abdul Rahman

in partial fulfillment of the requirements

for the degree of

BACHELOR OF INFORMATION SYSTEMS (HONOURS) INFORMATION SYSTEMS

ENGINEERING

Faculty of Information and Communication Technology

(Kampar Campus)

JAN 2023

Title:	DIET	[MANAGER
	Academic Session:	YEAR 3 SEMESTER 3
I	LEE	CHEE WAH
	(CAPITA	AL LETTER)
2. The Lib	brary is allowed to make copies	s of this dissertation for academic purposes.
2. The Lit	brary is allowed to make copies	s of this dissertation for academic purposes. Verified by,
2. The Lit	brary is allowed to make copies	s of this dissertation for academic purposes. Verified by,
2. The Lit	brary is allowed to make copies	s of this dissertation for academic purposes. Verified by, (Supervisor's signature)
2. The Lit (Author's s Address:	brary is allowed to make copies	s of this dissertation for academic purposes. Verified by,
2. The Lit (Author's s Address: 29, Lebuh F	brary is allowed to make copies	s of this dissertation for academic purposes. Verified by, (Supervisor's signature)
2. The Lit (Author's s Address: 29, Lebuh F Taman Ipoł	brary is allowed to make copies	s of this dissertation for academic purposes. Verified by, (Supervisor's signature) Abdulkarim Kanaan Jebna
2. The Lit (Author's s Address: 29, Lebuh F Taman Ipol 31400, Ipol	brary is allowed to make copies With the second se	s of this dissertation for academic purposes. Verified by, (Supervisor's signature) Abdulkarim Kanaan Jebna Supervisor's name

Universiti Tunku Abdul Rahman		
Form Title : Sample of Submission Sheet for FYP/Dissertation/Thesis		
Form Number: FM-IAD-004Rev No.: 0Effective Date: 21 JUNE 2011Page No.: 1 of 1		
FACULTY/INSTITUTE* OF Information and Communication Technology UNIVERSITI TUNKU ABDUL RAHMAN 		
SUBMISSION OF FINAL YEAR PROJECT /DISSERTATION/THESIS		
It is hereby certified that <u>Lee Chee Wah</u> (ID No: <u>1902956</u>) has completed this final year project/ dissertation/ thesis* entitled " <u>Diet Manager</u> " under the supervision of <u>Abdulkarim Kanaan Jebna</u> (Supervisor) from the Department of <u>Information and</u> Communication Technology_, Faculty of <u>Information and Communication Technology</u> .		
I understand that University will upload softcopy of my final year project / dissertation/ thesis* in pdf format into UTAR Institutional Repository, which may be made accessible to UTAR community and public.		
Yours truly,		
Unulubli (LEE CHEE WAH)		
*Delete whichever not applicable		

DECLARATION OF ORIGINALITY

I declare that this report entitled "**DIET MANAGER**" is my own work except as cited in the references. The report has not been accepted for any degree and is not being submitted concurrently in candidature for any degree or other award.

Unelible

Signature :

Name : LEE CHEE WAH

Date : 28/4/2023

ACKNOWLEDGEMENTS

I would like to express my deepest appreciation and profound gratitude to my esteemed supervisor, Abdulkarim Kanaan Jebna, for his unwavering support, invaluable guidance, and constant encouragement throughout the development of the Diet Manager project. As a newcomer to the world of mobile application development, his wealth of knowledge, expert advice, and constructive feedback have been instrumental in shaping my understanding and facilitating my growth in this domain.

Mr. Jebna's dedication to excellence, passion for innovation, and commitment to my personal and professional development have made a lasting impact on my academic journey. His tireless efforts to challenge and inspire me have not only fostered my growth as a mobile application developer but have also honed my critical thinking and problem-solving abilities.

I am truly fortunate to have the opportunity to learn from such an accomplished and distinguished mentor. I am profoundly grateful for his unwavering belief in my potential and his constant efforts to help me achieve it. My success in this project is a testament to his exceptional mentorship, and I am honored to express my sincere thanks to him for his invaluable contribution to my growth and development.

Thank you, Mr. Jebna, for your tremendous support and guidance throughout this project. I am forever grateful for the knowledge, skills, and confidence I have gained under your mentorship, and I look forward to carrying these with me as I embark on my future endeavors.

ABSTRACT

Mobile applications have become essential for individuals seeking to manage their health and lifestyle, particularly wellness apps that offer self-monitoring and diet tracking features. The project reviews on three of the current existing system, and critically analysed each of their advantages and limitation of the system. The Diet Manager app addresses issues in existing apps, such as lack of motivation and user disinterest, hence, the aim of the thesis is the implementation of better diet calories counting features with providing by implementing timelapse pictures of user progress and a streak system to reward consistency. This project aims to develop an Android app tailored for individuals with weight concerns, incorporating several sub-objectives: a diet management module, weight tracking tool, summary report system, reminder tool, timelapse feature, and streak system. The Phased Development methodology was used, adhering to the System Development Life Cycle (SDLC) for planning, analysis, design, implementation, and testing. The Diet Manager app provides a comprehensive solution for self-monitoring, diet tracking, and motivation enhancement, encouraging users to adopt and maintain a healthier lifestyle. Future improvements could include custom food item addition, personalized dietary recommendations, activity monitoring, and broader platform compatibility, enhancing user experience and increasing the app's utility in the health and wellness sector. In conclusion, the Diet Manager app offers a systematic, high-quality, and user-friendly solution for individuals seeking to manage their weight and improve their overall health.

TABLE OF CONTENTS

TITLE PAGE	Ι
REPORT STATUS DECLARATION FORM	Π
FYP THESIS SUBMISSION FORM	III
DECLARATION OF ORIGINALITY	IV
ACKNOWLEDGEMENTS	V
ABSTRACT	VI
TABLE OF CONTENTS	VII
LIST OF FIGURES	XI
LIST OF TABLES	XVI
CHAPTER 1 INTRODUCTION	1
1.1 Problem Statement	2
1.1.2 Motivation	4
1.2 Objectives	4
1.3 Project Scope and Direction	5
1.4 Contributions	5
1.5 Report Organization	6
CHAPTER 2 LITERATURE REVIEW	9
2.1 Previous Works on Diet Manager	9
2.1.1 Lose it!	9
2.1.2 FatSecret	13
2.1.3 MyNetDiary	17
2.2 Limitations of Previous Studies	21
2.2.1 Lose it!	21
2.2.2 FatSecret	21
2.2.3 MyNetDiary	21
CHAPTER 3 SYSTEM METHODOLOGY/APPROACH	23

3.1 System Methodology

Bachelor of Information Systems (Honours) Information Systems Engineering Faculty of Information and Communication Technology (Kampar Campus), UTAR 23

3.1.1 Planning Phase	23
3.1.2 Design Phase	24
3.1.3 Implementation Phase	24
3.1.4 Testing Phase	24
3.1.5 Phased Development	25
3.2 System Design Diagram	26
3.2.1 Use Case Diagram	26
3.2.2 Activity Diagram	27
3.3 Project Time	36
3.3.1 Gantt Chart	36

CHAPTER 4 SYSTEM DESIGN	38
4.1 Firebase Firestore Tree-Like Structure	38
4.2 Firebase Realtime Database Tree-Like Structure	39
4.3 Sitemap of Diet Manager	41
4.4 System UI Design	43
4.4.2 UI Design of Record Page screen	43
4.4.3 UI Design of Progress Page screen	45
4.4.4 UI Design of Reminder Page screen	46
4.4.5 UI Design of Timelapse Page screen	47
4.4.6 UI Design of Streak Page screen	48
4.4.7 UI Design of User Profile Page screen	49
4.5 Client Server Architectural Design	51

CHAPTER 5 SYSTEM IMPLEMENTATION	53
5.1 Hardware Setup	53
5.2 Software Setup	55
5.2.1 Flutter Set-up	55
5.2.2 Java Set-up	55
5.2.3 Android Studio Set-up	55
5.2.4 Firebase Set-up	55
5.2.5 Mobile Device Set-up	56

5.3 Setting and Configurations	57
5.3.1 Flutter Set-up	57
5.3.2 Java Set-up	59
5.3.3 Android Studio Set-up	62
5.3.4 Firebase Set-up	66
5.3.5 Mobile Device Set-up	81
5.4 System Operations	83
5.4.1 User Authentication	83
5.4.2 Page Navigation	87
5.4.3 Record Page	89
5.4.4 Progress Page	96
5.4.5 Welcome Page	100
5.4.6 Reminder Page	104
5.4.7 Timelapse Page	107
5.4.8 Streak Page	112
5.4.9 User Profile Page	117
5.5 Implementation Issues and Challenges	122
5.6 Concluding Remarks	123

CHAPTER 6 SYSTEM EVALUATION AND DISCUSSION	125
6.1 System Testing and Performance Metrics	125
6.2 Testing Setup and Result	126
6.2.1 Record Page	127
6.2.2 Progress Page	129
6.2.3 Reminder Page	132
6.2.4 Timelapse Page	133
6.2.5 Streak Page	135
6.2.6 User Profile Page	136
6.3 Project Challenges	138
6.4 Objectives Evaluation	140
6.4.1 Evaluation of Sub-objective 1	140
6.4.2 Evaluation of Sub-objective 2	140
6.4.3 Evaluation of Sub-objective 3	140

6.4.4 Evaluation of Sub-objective 4	140
6.4.5 Evaluation of Sub-objective 5	141
6.4.6 Evaluation of Sub-objective 6	141
6.5 Concluding Remark	142

CHAPTER 7 CONCLUSION AND RECOMMENDATION	143
7.1 Conclusion	143
7.2 Recommendation	144

REFERENCES	145
WEEKLY LOG	146
POSTER	154
PLAGIARISM CHECK RESULT	155
FYP2 CHECKLIST	161

LIST OF FIGURES

Figure Title	Page
Figure 2. 1 The setup page for user	11
Figure 2. 2 The user basic information	11
Figure 2. 3 The user calculated weight plan	11
Figure 2. 4 The navigation tab of the app	11
Figure 2. 5 The search food function of the app	11
Figure 2. 6 The Log tab in the app	11
Figure 2. 7 The Goals tab in the app	12
Figure 2. 8 The My Day tab in the app	12
Figure 2. 9 The Social tab in the app	12
Figure 2. 10 The Reminder functions in the app	12
Figure 2. 11 The question s for user to set up their diet plan	15
Figure 2. 12 The recommended diet plan set up by the app	15
Figure 2. 13 The Diary tab of the app	15
Figure 2. 14 The search function of the app	15
Figure 2. 15 The Report tab of the app	15
Figure 2. 16 The Macro section of the Report tab in the app	15
Figure 2. 17 The Nutrients section of the Report tab in the app	16
Figure 2. 18 The Weight tab of the app	16
Figure 2. 19 The Home tab of the app	16
Figure 2. 20 The reminder function of the app	16
Figure 2. 21 The Diet Calendar planner for the app	16
Figure 2. 22 The app questing user last meal	19
Figure 2. 23 The app have search function	19
Figure 2. 24 The Dashboard tab app show calories intake for every meal	19
Figure 2. 25 The Dashboard tab app show daily calories budget	19
Figure 2. 26 The Coach Tab at the app	19
Figure 2. 27 The Diets Tab at the app	19
Figure 2. 28 The premium functions of the app	20
Figure 2. 29 The Me tab of the app	20
Figure 2. 30 The set reminder function of the app	20

Figure 2. 31 The set reminder function of the app	20
Figure 3. 1 Use Case Diagram	26
Figure 3. 2 Calculate the calories intake function	27
Figure 3. 3 Record diet progress function	29
Figure 3. 4 Read summary report function	30
Figure 3. 5 Remind user input calorie intake function	32
Figure 3. 6 Create a timelapse function	33
Figure 3. 7 Collect streak function	35
Figure 3. 8 Gantt Chart for the First Version of the app in the first semester	36
Figure 3. 9 Gantt Chart for the Second Version of the app in the second semester (part
1)	37
Figure 3. 10 Gantt Chart for the Second Version of the app in the second semester (part
2)	37
Figure 4. 1 Firebase Firestore Tree-Like	38
Figure 4. 2 Firebase Firestore Tree-Like	39
Figure 4. 3 Sitemap of Diet Manager	41
Figure 4. 4 UI Design of Record Page	43
Figure 4. 5 UI Design of Food History List Page from Record Page	44
Figure 4. 6 UI Design of Food History Details Page from Food History List Page	44
Figure 4. 7 UI Design of Progress Page	45
Figure 4. 8 UI Design of Reminder Page	46
Figure 4. 9 UI Design of Timelapse Page	47
Figure 4. 10 UI Design of Streak Page	48
Figure 4. 11 UI Design of User Profile Page	49
Figure 4. 12 Client Server Architectural Design of the Diet Manager application	51
Figure 5. 1 Search "environment variables" in the start menu	57
Figure 5. 2 System Properties Window	58
Figure 5. 3 Environmental Variables Window	58
Figure 5. 4 Edit environment variable Window	59
Figure 5. 5 Search "environment variable" in the start menu	60
Figure 5. 6 System Properties Window	60
Figure 5. 7 Environment Variables Window	61
Figure 5. 8 Edit User Variable Window	61
Figure 5. 9 Search Flutter in the Plugins tab of Welcome to Android Studio Screen	62

Figure 5. 10 Search Dart in the Plugins tab of Welcome to Android Studio Screen	63
Figure 5. 11 Type "flutter doctor - android-licenses." in Command Prompt Screen	63
Figure 5. 12 Type "flutter doctor" in Command Prompt Screen	64
Figure 5. 13 SDK Tools Screen in Android Studio Settings window	64
Figure 5. 14 Open files in the Project screen in Android Studio	65
Figure 5. 15 Add Project screen on the Firebase website	66
Figure 5. 16 Add Project Screen 2 to the Firebase website	66
Figure 5. 17 Firebase main page	67
Figure 5. 18 Firebase Authentication page	68
Figure 5. 19 Firebase Authentication Sign-in providers page	68
Figure 5. 20 Firebase Authentication Sign-in Provider Status	69
Figure 5. 21 Cloud Firestore Create database page	69
Figure 5. 22 Cloud Firestore Create database page 2	70
Figure 5. 23 Indexes Tab on the Cloud Firestore page	70
Figure 5. 24 Composite Index Details page	71
Figure 5. 25 Firebase Realtime Database Set up database page	72
Figure 5. 26 Firebase Realtime Database Set up database page 2	72
Figure 5. 27 Rules Tab on the Realtime Database page	73
Figure 5. 28 Firebase Set up Cloud Storage page	73
Figure 5. 29 Firebase Set up Cloud Storage page 2	74
Figure 5. 30 Project settings in the Project Overview tab of the Firebase project p	bage
	75
Figure 5. 31 Your apps section in General Tab	76
Figure 5. 32 Add Android app page Step 1	77
Figure 5. 33 Add Android app page Step 2	78
Figure 5. 34 google-services.json location in the Android Studio project file direct	tory
	79
Figure 5. 35 Add Android app page Step 3 (part 1)	80
Figure 5. 36 Add Android app page Step 3 (part 2)	80
Figure 5. 37 Developer options screen in mobile device's Settings	81
Figure 5. 38 USB Settings screen	82
Figure 5. 39 Connected device at the top of the screen of Android Studio	82
Figure 5. 40 Diet Manager's Sign In screen	84
Figure 5. 41 Diet Manager's Sign Up screen	85

Figure 5. 42 Diet Manager's Reset Password screen	86
Figure 5. 43 Reset Password email in Inbox	86
Figure 5. 44 Reset Password webpage	87
Figure 5. 45 Diet Manager's Bottom Navigation Bar	88
Figure 5. 46 Diet Manager's Profile Icon	88
Figure 5. 47 Diet Manager's Navigation Drawer	89
Figure 5. 48 Diet Manager's Record Page	90
Figure 5. 49 Add food item dialogue box	91
Figure 5. 50 Calories values in Add food item dialogue box	92
Figure 5. 51 Quantity values in Add food item dialogue box	92
Figure 5. 52 Food items listed food list	93
Figure 5. 53 Edit food item dialog box	93
Figure 5. 54 Record Page top half screen when calories are within the limit	94
Figure 5. 55 Record Page top half screen when calories exceed the limit	94
Figure 5. 56 Food History List screen	95
Figure 5. 57 Food History Details screen	96
Figure 5. 58 Diet Manager's Progress Screen	97
Figure 5. 59 Add a new weight dialogue box	98
Figure 5. 60 Weight item list in Progress Screen bottom half screen	99
Figure 5. 61 Swiping weight items	99
Figure 5. 62 The top half screen of the Progress Page Screen	100
Figure 5. 63 Diet Manager's Welcome Page	101
Figure 5. 64 Harris–Benedict equation slide screen	102
Figure 5. 65 TDEE slide screen	103
Figure 5. 66 Create Calories Deficit slide screen	104
Figure 5. 67 Diet Manager's Reminder screen	105
Figure 5. 68 Time Picker dialogue box	106
Figure 5. 69 Notification On Toggle button	106
Figure 5. 70 Reminder Notification	106
Figure 5. 71 Diet Manager's Timelapse screen	107
Figure 5. 72 Timelapse screen when pictures selected	108
Figure 5. 73 Built-in camera function screen	108
Figure 5. 74 Pictures view in Fullscreen screen	109
Figure 5. 75 Files Permission request when downloading Pictures to phone st	orage110

Figure 5. 76 Files Permission request when creating and saving	Timelapse video to
phone storage	111
Figure 5. 77 Created Timelapse video file	111
Figure 5. 78 Diet Manager's Streak screen	112
Figure 5. 79 Redeem Streak dialog box	113
Figure 5. 80 Streak count displayed on Streak screen	114
Figure 5. 81 Redeemed Failed dialog box	115
Figure 5. 82 Streak Reduced dialog box	116
Figure 5. 83 Calories Exceeded Failed dialog box	117
Figure 5. 84 Diet Manager's User Details screen	118
Figure 5. 85 SMART Diet Goal dialog	119
Figure 5. 86 User Profile's user details used for calculation	120
Figure 5. 87 User Profile's Save Icon button	120

LIST OF TABLES

Table	Title	Page
Table 5.	1 Record Page Screen and Details	96
Table 5.	2 Progress Page Screen and Details	100
Table 5.	3 Welcome Page Screen and Details	104
Table 5.	4 Reminder Page Screen and Details	106
Table 5.	5 Timelaspe Page Screen and Details	111
Table 5.	6 Streak Page Screen and Details	117
Table 5.	7 User Profile Page Screen and Details	121
Table 6.	1 Integration Testing for Record Page Functions	128
Table 6.	2 Integration Testing for Progress Page Functions	131
Table 6.	4 Integration Testing for Reminder Page Functions	132
Table 6.	5 Integration Testing for Timelapse Page Functions	134
Table 6.	6 Integration Testing for Streak Page Functions	135
Table 6.	7 Integration Testing for User Profile Page Functions	136

CHAPTER 1 INTRODUCTION

With the technological advancements today, mobile application in our mobile devices has increasingly become more and more popular with tools that could ease our daily lives. Mobile applications these days evolved into tools that help individuals take control of their lifestyle and health, especially wellness applications where individuals want to perform self-monitoring and diet tracking with the functions and guidance such as calorie-tracking functions provided by the app. A study showed that more than half of the people in the United States have downloaded at least one health-related app [1]. Part of the motivation for this project is to contribute to the wellness application sector to help users to improve their healthy lifestyles. In this chapter, we present the background and motivation of our research and the outline of the thesis.

Wellness mobile application can be utilized to keep track of users' health and assist them in keeping a healthy lifestyle; it plays a significant role in keeping track of and regulating an individual's weight [1]. The diet managing apps falls under the category of wellness apps. A diet manager app, as its name implies, is a weight management app for users to monitor their diet by performing self-monitoring on themselves. The main functions of these apps include calculating daily food calorie intake, setting targeted weight goals, tracking user daily calorie intake and tracking the progress of the user's diet performance.

The purpose of this proposal is to introduce the background, innovation, problem statement of the topic, motivation of the research, the project scope and objectives of the proposal, the contribution to the field, and the outline of the thesis. The proposal will explain the problem of the diet manager app, which describes the problem of the app requiring users constantly monitor food intake, the lack of explanation of diet apps for users to understand the dieting strategies involved, and the issue of users forgetting to record their meals in the apps. The motivation of the research describes the reasons behind the implementation of better diet calorie counting features by providing detailed guides for users about dieting. Besides that, the scope of the project describes the implementation of a diet managing system into an Android mobile platform for Android users. The contribution describes the implementation of diet management.

1.1 Problem Statement

i. Users may lose interest or give up using the app when results do not meet their expectations.

A dieting app is a tool that user input their efforts to start their dieting journey, and the output would be their desired weight change of the user. But if the results do not favour the user, the user may lose the motivation to continue dieting and may eventually lead to stop using the app. The diet app lacks the features to create something that is meaningful to the user, something that makes the user think that all the effort that the user put into dieting is meaningful in some ways, even if their results do not favour the user.

According to a study conducted [2], the study sought to better understand user views and experiences in order to inform the development of future smartphone applications for health behaviour change. The research identified numerous characteristics that influenced user engagement with the apps. Participants emphasised the importance of personal relevance, which means that the app should be adapted to individual needs and tastes. They also emphasised the importance of user-friendliness, with intuitive interfaces and simple navigation. Users also praised the apps' capacity to provide reliable and accurate information. Social support options, such as discussing progress with friends or joining communities with similar goals, were also deemed beneficial for motivation and engagement.

Such a problem can be solved by implementing a timelapse feature of the user's diet progress. The timelapse feature collects and compiles the pictures taken by users daily throughout their diet progress and makes it into a timelapse of pictures displaying pictures of the user from the beginning of the diet journey until the end of the diet journey. The timelapse can be downloaded and shared with friends. The user can create something about themselves going through a diet journey from the app. Users will find the diet journey that they went through may not necessarily be pointless, and the timelapse that they create using the diet journey can keep them motivated and get back to using the app and continue to diet.

ii. The user lacks the motivation to lose weight.

Dieting app requires the user's constant effort to monitor their food consumption and daily activities routine, but because of this, many users drop off quickly after one or two months of the initial usage of the app [2]. The possible reason behind this can be that users' only interest in using diet apps is because they want to lose or gain weight. If the user losses interest in dieting, they will stop using the app. The diet app lacks certain features to keep the continuous motivation the user had when they first started to use the diet app to diet. Such a problem can be solved by implementing a streak system that can be used to keep the user motivated to use the diet app. Users will collect one streak point for each day they manage to keep their food calorie intake within the limit throughout their dieting journey. As each dieting day goes by, let's say 70 days, the streak points increase up to 70 points, and the user gets to keep their streak. If users missed to collect any of their streak points for the day, they would lose their streak, which means the streak points will be down to 0, and they will have to start all over again. The number of streak points that are accumulated can be used to display or symbolize the amount of hardship and determinacy of a person, and the number can be shared with friends. User will try not to lose their streak because they might think all the effort they put into collecting the streak points daily will all goes to waste. This situation may create a habit-forming to keep users using the app, and users will engage more with the app.

In short, there is no good application able to provide the convenience of user input food intakes, detailed explanation on the diet calculation as well as the confidence of the user to app for their time commitment. Thus, there is an agile need for such an application to assist in providing user diet information that users can fully understand and use. The diet manager will provide all the basic diet managing functions such as calories tracking module, weight tracking tool, record and estimate time required to reach weight goal, summary report system to monitor to keep track of diet goal, reminder tool to input daily calories, as well as functions that can keep the interest of the user to continue using the app that other diet app did not have, such as a function to create timelapse of user diet progress and function to count user streak.

1.1.2 Motivation

With the fast-growing development of technology, especially in the mobile application development sector, mobile application not only provides humans with productive functions such as basic calling and messaging or email functions, but they also evolved into tools that help to improve human lifestyle. This is especially true when mobile applications these days provide humans with personal needs such as a self-managing diet. Integrating a diet managing system into the mobile application can supply the needs of individuals who want to manage their weight, and a mobile application is preferable and welcomed by individuals compared to hiring a personal health fitness instructor or attending fitness classes. Hence, the aim of the thesis is the implementation of better diet calorie counting features by providing a timelapse system to create an overview of the user's diet progress using pictures and a streak system to keep the user motivated to use the diet app by having user building streak in the app, to solve the problems of the existing system mentioned.

1.2 Objectives

Main objective: To develop a diet manager system for individuals with weight concerns by implementing the system into an Android mobile application.

Sub-objectives:

- 1. Develop a diet management module to calculate the user's food calorie intake in every meal.
- 2. Develop a weight tracking tool to record users' weight and for users to keep track of their weight.
- 3. Develop a summary report system to monitor user dieting progress to keep track of user diet goals.
- 4. Develop a reminder tool for users to input user's food calorie intake when the users forget to.
- 5. Develop a timelapse of the user's diet progress using pictures that the user saves daily for the user to have an overview of their diet progress.
- 6. Develop a streak system for the user to collect streak points when the user does not exceed their daily calorie intake limit.

1.3 Project Scope and Direction

The expected outcome of this project is a piece of android mobile application software with the diet manager system implemented. The diet manager system is targeted to android mobile phone users with weight concerns looking for an application to monitor their diet.

1.4 Contributions

The contribution of this project is to develop a diet managing system into an Android mobile application with new, improved features such as:

- 1. Provide a tool for users to create a timelapse of their diet progress by collecting and compiling all the pictures taken by the user daily throughout their dieting journey. The diet manager app must provide a meaningful experience for the user when the user diet. Most diet apps only provide the dieting functions for the user to use, and the output of their dieting journey results is their weight change. But not all dieting results are in favour of the user, and some results may not come out as the user expected. This may create some demotivation or loss of interest for the user to continue their dieting journey, but this is where the timelapse function comes and makes the whole dieting experience more pleasant and meaningful. The user can take a picture of themselves daily throughout their dieting journey, and the app will collect and compile all the pictures in sequence from the beginning of the dieting day towards the end of the dieting day to create a wonderful timelapse of pictures of their body changes. This timelapse can be downloaded into a file and shared with people. This can create a sense of success to congratulate the user for their achievement from the app itself and also creates a sense that the diet journey is not pointless for the user that their weight goal is not achieved and may motivate the user to continue their effort in dieting.
- 2. Provide an interesting streak system whereby the user continues to redeem daily streak points and does not break the streak whenever the user does not overconsume their daily calorie limit. The diet manager app provides a streak system whereby users can collect a streak point daily in the app if they do not over-consume their daily calorie intake limit throughout their dieting journey. The streak will break if the user does not collect the streak point or over-consumes their daily calorie intake. For example, when a user has been collecting the streak for more than six weeks, which is equivalent to 42 streaks, it creates the urge for the user to keep the streak

for all the effort the user put into accumulating that particular amount of streak points This creates the motivation for the user to continue on diet dieting journey, as well as making their dieting journey experience fun. It may also make use to have the continuous use of the app, and as time goes by, the continuous effort will slowly become a daily habit for the user to use the app and record their daily calorie intake.

1.5 Report Organization

The content of this project is displayed in the following five chapters. In Chapter 1, describe the project background, as well as the three problems that the currently existing system has, followed by the motivation of the project, which is to implementation of better diet calorie counting features by providing detailed guides to educate users about dieting, to solve the problems of the existing system mentioned. Then, it also describes four project objectives and the two novelties that contributed to developing the system for the project.

Then, Chapter 2 describes the three previous works on diet manager, which are Lose it!, FatSecret, and MyNetDiary, as well as followed by the limitation for each system faced.

Then, Chapter 3 describes the System Methodology used for the development of this project which is the SDLC Waterwall approach. And the chapter is followed by the system design diagram, which is the use case diagram and the activity diagram. And lastly And lastly the chapter ends with the project timeline, which discusses the duration required for the entire development of the Diet Manager will be divided into two semesters, with each version of the app for each semester to develop.

Then, Chapter 4 describes the System Implementation. And then, the subchapter within it describes the Firestore Database Tree-Like diagram and Firebase Realtime Database design which is the database model used for the Diet Manager project. Then the chapter follows with a Sitemap diagram, which discusses the Diet Manger's structure and navigations between pages and functions. Besides that, it also described the UI design of all the pages screen that contains the main functions of the Diet Manager. Lastly, it also discussed the architecture design used for the Diet Manager, which is the Client Server Architectural Diagram.

Then, Chapter 5 describes the System Design and Implementation. The following chapter describes the Hardware Setup required, such as devices used for the implementation of the Diet Manager. Then followed by a brief discussion of the

components of the Software Setup required, such as the Flutter Set-up, Java Set-up, Android Studio Set-up, Firebase Set-up and Mobile Devices Set-up for the implementation of the Diet Manager. Then followed by the discussion of more elaborated and detailed Settings and Configuration required for each of the Software Setup components mentioned previously. Then, followed by the System Operations, which discussed all available operations of each page of the Diet Manager, such as the User Authentication, Page Navigations, Record page, Progress Page, Welcome page, Reminder Page, Timelapse Page, Streak Page and User Profile Page. Then the chapter followed by a discussion on the Implementation Issues and Challenges mentioned, such as user-friendly and intuitive UI/UX, integration with Firebase and Calculation accuracy. And lastly, the chapter ends with the concluding remark, which briefly summarizes all work that is discussed in Chapter 5.

Then Chapter 6 describes the System Evaluation and Discussion. The chapter begins with the system testing and performance metrics, which provide an overview of the system testing performed on the application, focusing on integration testing techniques and their role in ensuring seamless functionality between different components. And it describes the performance metrics used to evaluate the effectiveness of the testing process. Then in the following chapter, discussed the Integration testing was performed on only the major functions of the Diet Managers, which are the Record page, Progress Page, Reminder Page, Timelapse Page, Streak Page and one other function, User Profile Page. And the result of each test performed on each of the functions will be summarized. Then, The following chapter describes some of the project challenges, which are the UI design and Optimizing Performance. Then the chapter followed by the objectives evaluation, which evaluated all project objectives of the project met. And lastly, the chapter ends with the concluding remark, which briefly summarizes all work that is discussed in Chapter 6.

Then Chapter 7 described the Conclusion, which summarized the work done in the whole project. Then the chapter followed by the recommendations which introduced some of the limitations which were difficult for users to add unique food items to their diet plans and the limited tracking of physical which may not track this crucial component of leading a healthy lifestyle of the project and the recommended solutions included a function that lets users design their own food products and include them in their diet plan to get around this restriction and include a feature that enables users to Bachelor of Information Systems (Honours) Information Systems Engineering Faculty of Information and Communication Technology (Kampar Campus), UTAR manually track their physical activity or interact with fitness trackers to overcome this issue.

CHAPTER 2 LITERATURE REVIEW

This chapter review the previous works on diet management applications available on mobile platforms, focusing on their features, functionalities, and limitations. By examining the strengths and weaknesses of existing diet management applications, this literature review aims to identify potential areas of improvement that can enhance the overall user experience and effectiveness of these apps in helping users reach their weight management goals. This chapter begins with an overview of three popular diet management applications, namely Lose it!, FatSecret, and MyNetDiary, discussing their key features, functionalities, and user interfaces. The the chapter followed by the limitations of these applications, highlighting potential areas for improvement. Lastly, the subsequent sections discussed the details of each application and examine their specific limitations.

2.1 Previous Works on Diet Manager

2.1.1 Lose it!

Lose it! is a diet manager application available on the mobile Android platform. The app was founded in 2008 with the aim of making counting calories simpler. The main objective of the creation of the app is to help individuals to reach a healthy weight by providing them with the proper tools to reach their goals. The app is based on three main key aspects, set goals, track food, and lose weight.

When opening the app, the app will walk through new users with specified steps to help user to set up their weight loss success (Refer to Figure 2.1). There are six steps for the user to set up their profile. Most of the step only requires the user to answer questions and collect feedback. However, we are more interested in reviewing the main functionality of the app, and we will skip between important steps related to the main functions. We skip to the second step, where the user will input their information such as current birthdate, height, weight and gender (Refer to Figure 2.2). Then in the fourth step, the user will input their goal weight and how quickly the user wants to reach the goals. The app will calculate and estimate the daily calorie cap and time for the user to reach their target goal (Refer to Figure 2.3). After the user walkthrough all the steps, the app will lead to the main page of the app. There are four tabs at the bottom of the screen for the user to navigate between them (Refer to Figure 2.4).

We will start with the first tab, which is the Log tab. The Log enables the user to input each meal the user takes in. The app has a search function to browse all types of food and display the calories for each of them (Refer to Figure 2.5). The top part of the page shows the maximum number of daily calories for the user to keep track of (Refer to Figure 2.6).

The next tab is the Goals tab. The Goals tab shows the current progress of the user's diet. It keeps track of the user's weight by daily by having user to enter their weight occasionally. The app will make adjustment according and estimate the number of days left for the user to reach their goals (Refer Figure 2.7).

The next tab which is the My Day tab. The My Day tab shows a simple summary of the daily and weekly calories intake by the user. It also shows the daily streak of the user following according to the recommend calories intake (Refer Figure 2.8).

The last tab is the Social tab. The social tab allows users to post their thoughts and messages in the community for other users that uses the app to see. The user can add friends and message them using the app (Refer Figure 2.9).

The apps have the features for user to set reminder for user to input their calories intake in case the user forgets (Refer Figure 2.10).

The strength of Lose it! are that it the user interface is very easy for user to view and understand. For example, the main tab which is Log tab shows all the information the user needs to keep track of their calorie intake which makes it very straightforward and very easy for the user's eye to navigate the content instantly. There are only four simple tabs for user to navigate, which the app simple to use for all ages, especially to users that are not familiar with using mobile phone. The app also has a broad directory of all types of food information and its calories for user easily input their food calories in each meal without the user to enter the calories of the food taken themselves.



Figure 2. 1 The setup page for user

Figure 2. 2 The user basic information

Figure 2. 3 The user calculated weight plan



Figure 2. 4 The navigation tab of the app

Figure 2. 5 The search food function of the app

Figure 2. 6 The Log tab in the app



Figure 2. 7 The Goals tab

```
in the app
```

Figure 2.8 The My Day tab

in the app

in the app

Figure 2. 9 The Social tab



Figure 2. 10 The Reminder functions in the app

2.1.2 FatSecret

FatSecret is a diet managing app available for Android mobile platform. The app was founded in 2007 in Melbourne, Australia by Lenner Moses and Rodney Moses. The main objective of this app is for calorie counting and meal tracking.

When opening the app, the app will walk through the user with few questions to help user to calculate the recommended daily calories and set up the profile (Refer Figure 2.11). The question will have user to answer their targeted goals, gender, activity level, current weight, height, birthdate, region and how fast the user wants to achieve their goal. The app will use the information gathered by the user and calculate the recommended daily calories intake and the estimated time required for the user to lose weight (Refer Figure 2.12). After setting up the user profile, the app led the user to their home page in the app. There are 4 tabs at the bottom of the app for user to navigate between tabs.

The first tab is Diary tab. The Diary tab displays the daily recommended calories intake for user to follow (Refer Figure 2.13). The apps allow user to enter the food they consumed for each meal. The apps allow user to input their meal by searching their food in their app (Refer Figure 2.14).

The next tab is Reports tab. The Reports tab displays the daily and weekly calories intake of the user in the form of bar graph (Refer Figure 2.15). There are other sections in this page as well, such as Macros that display the nutrients of the food taken by the user for the day (Refer Figure 2.16). The other sections which is the Nutrients, show more detailed version of the nutrients of the food taken by the user for the day (Refer Figure 2.17).

The next tab is Weight tab. The Weight tab tracks the weight of the user daily, by having user to input their weight in the app occasionally. The app will keep a record of the weight daily and display the amount of weight left to reach the targeted goal (Refer Figure 2.18).

The last tab is Home tab. The Home tab shows the diet progress of other users that using the app (Refer Figure 2.19). The user can share their progress in the app for other users to see as well.

The app also has the options to set reminder for user to do task (Refer Figure 2.20). User can set the reminder to weight themselves, or to input the food taken in each meal to remind user in case they forget.

The strength of the FatSecret app is that it provides user with very detailed information such as the nutrients of the food taken by the user for the day. The app also has a wide and huge database of types of food and its calories for user to search and input their food intake for each meal. The app has a simple UI which makes the content easy for user to navigate around the app. The app also has diet calendar where it allows user to record the activities performed by the user everyday (Refer Figure 2.21).



Figure 2. 11 The question s for user to set up their diet plan

× Breakfast ▼ Tuesday, Nov 30	Ø
COOK BOOK FOOD RECENTLY E	ATEN
X Q eggs	8
Egg 1 large RDI 8% - 74 kcal	
Egg White 1 large RDI 2% - 17 kcal	
Boiled Egg 1 large RDI 9% - 77 kcal	
Scrambled Egg (Whole, Cooked) 1 large RDI 11% - 101 kcal	
Fried Egg 1 large RDI 10% - 92 kcal	
Poached Egg 1 large RDI 8% - 73 kcal	
Egg Yolk 1 large RDI 6% - 55 kcal	
Egg Omelet or Scrambled Egg 1 large egg RDI 11% - 98 kcal	
Soft Boiled Egg 1 medium RDI 8% - 68 kcal	
Egg Substitute (Liquid) 1 cup RDI 23% - 211 kcal	
Fried Eag without Fat	_

Figure 2. 14 The search function of the app

Figure 2. 12 The recommended diet plan set up by the app

≡	Repor	ts				ø
	This Wee	ek				▼ ▶
CALC	DRIES	2	MACR	os	N	JTRIENTS
Calorie 441 Daily A	s verage: 44	41			Goal	: 900kcal
Mo 29	Tu 30					Su 5 Cals
Brea	akfast				(33%)	(kcal) 147
Lune					(33%)	
Dinr	ner				(33%)	147
Sna	cks/Other				(0%)	
Food	s Eater	ı			Times	Cals
home	•	Diary		Report		Weight

Figure 2. 15 The Report tab of the app

Figure 2. 13 The Diary tab

of the app Reports ø ≡ This Week 4 MACROS CALORIES NUTRIENTS Macronutrients Goa Total Carbohydrate Fat 30% Protein 20% Foods Eaten Fat (g) Carbs (g) Prot (g)

Figure 2. 16 The Macro section of the Report tab in the app

Egg

Total

n

2.31 29.82 37.74

2.31 29.82 37.74

ıh

\equiv Reports			ø
◀ This Week		,	-
CALORIES	MACROS	NU [.]	TRIENTS
Nutrients			
Nutrient	Total	Goal	[+/-]
Calories (kcal)	441	6300	-5859
Protein (g)	38	700	-662
Carbohydrate (g)		1750	-1748
Fiber (g)			-
Sugar (g)			-
Fat (g)	30	469	-439
Saturated Fat (g)			-
Polyunsaturated Fat	t (g) 4		÷
Monounsaturated Fa	at (g) 11		-
Cholesterol (mg)	1269		-
Sodium (mg)	420		-
Home Di	ary Reports	5	Weight



Figure 2. 18 The Weight tab of the app



Figure 2. 19 The Home tab of the app





Figure 2. 20 The reminder function of the app

November 2021 -441 30 ۳f Ŕ. + 29 + + 28 27 Sat + 26 ۳ſ 25 24 23 + 22 Mon 21 + + 20 + 19 + +

Figure 2. 21 The Diet Calendar planner for the app

2.1.3 MyNetDiary

MyNetDiary is an online wellness application available for both Android and iOS platform. The app was launched in 2007. The application main purpose served as calories and exercise tracker.

When opening the app, it immediately questioning the last meal the user had (Refer Figure 2.22). Then the app led the user to enter the food they had of the last meal. The app has the search functions to search all type of food and shows its calories for each food (Refer Figure 2.23). After adding the last meal, the app will record calories of the food for that meal and led user to the main page which is the Dashboard (Refer Figure 2.24).

The app has four tabs at the bottom of the screen for user to navigate between them. The first tab is the Dashboard tab. It displays the calories intake for each meal of the day and user create the calorie budget plan at the page. It begins setting up user profile by asking question such as gender, height, birthdate, current weight, and target weight goal. After setting up, the app will recommend the calorie budget for the user to follow, and the app will track the calories intake left for the day (Refer Figure 2.25). Besides that, the page also shows the total nutrients of the food user consume for the day.

The next tab is Coach tab. The Couch tab displays and tracks the progress of the diet. It provides the estimation of number of days required for the to reach their target weight goal (Refer Figure 2.26).

The next tab is Diet tab. The Diet tab shows different types of diet plan for user to choose from (Refer Figure 2.27). When user select the diet plan, it will explain to the user about the selected diet plan, but the features is only available for paid users with premium plan (Refer Figure 2.28).

The last tab is the Me tab. The Me tab shows the other functions of the app such as Recipe Import, for user to import recipe from external sources, Recipe Database for user to search food information, Measurements for user to save their blood pressure and Personal Info for user to update their profile information (Refer Figure 2.29).

The app also has the functions to Set Reminders for user to record their food every meal in case they forget (Refer Figure 2.30). The app also has functions to link the app with fitness trackers devices (Refer Figure 2.31).

The strength of the app is that it is very straightforward and easy to use. As an example, all the main functions such as calories tracking and estimation of number of days required to reached targeted goals displayed in simple way where user can understand immediately. The app also has broad database of many types of food for user to searched when recording their meals which makes it easy for user to obtain information of the food.



questing user last meal

۰ ک



Figure 2. 23 The app have search function



Figure 2. 24 The Dashboard tab app show calories intake for every meal



Figure 2. 27 The Diets Tab at the app

Exercise	Calorie Budget 1,339	Breakfast
0		0
Steps		Lunch
0 %	180	0
Water	Left	Dinner
00	1,159	180
Notes		Snacks
U	View All Meals	0
Carbs 2% cals	Protein 29% 13g left 54g	Fat 69% 14g left 38g
Log all da your aver	ays to get the best age intake.	picture of
Please lo	og more than 400 f	ood calories
to get we	eight forecast and	diet tips.
to get we	hight forecast and	diet tips.
to get we	light forecast and D	AILY ANALYSIS
To get we	light forecast and p	AILY ANALYSIS
Vutrients Sodium	190mg, 8% le	AILY ANALYSIS

Figure

2.

Dashboard tab app show

25



Figure 2. 26 The Coach Tab at the app

The

daily calories budget
CHAPTER 2



Figure 2. 28 The premium Figure 2. 29 The Me tab of functions of the app

=	Apps & Devices 🛛 👋 😰
•	Google Fit App Link Foods, Activities, Steps and Weight.
P	Samsung Health App Link Foods, Water, Steps and Weight
0	Twitter Tweet your progress and success
	Fitbit Activity Trackers Links: Calories Burned, Steps, Floors, Sleep Hours, Heart Rate, and Food. Supports: all models of Fitbit trackers and watches.
C	Garmin Activity Trackers Links: Exercise and Daily Steps Calories, Steps, and Sleep Hours. Supports: all models of Garmin watches and trackers that work with Garmin Connect.
•	Withings Activity Trackers Links: Daily Calories, Exercise, and Steps. Supports: all models of Withings watches and trackers.
-	Withings Weight Scale
0	Fitbit Aria Weight Scale

reminder function of the

app

Bachelor of Information Systems (Honours) Information Systems Engineering Faculty of Information and Communication Technology (Kampar Campus), UTAR

the app

Figure 2. 30 The set reminder function of the app

2.2 Limitations of Previous Studies

2.2.1 Lose it!

The app did not show the estimated days left for the user to achieve the targeted weight goal. Besides that, the app should also include features that are interesting that can create continuous user interest using the app, such as users can create something meaningful from the app, for example, a timelapse of pictures of themselves and collecting streak points to keep the streak throughout their dieting progress.

2.2.2 FatSecret

The summary report in the app does not estimate the days for the user to reach their targeted goals. Instead, it only shows the remaining weight for the user to lose. I find this information to be very less helpful, especially when users are eager to check if they can see results and if they can shorten the time required to lose weight if they lose weight faster than what is recommended by the app. The app should include and update the estimated dates or days left to reach their targeted goals every time the user updates their body weight to the app. Motivation is an important aspect that should be implemented in diet apps for users to stay motivated, for example, by providing realtime feedback and digital personal instruction [3]. Beside that, the app does not explain how calorie counting works in managing weight loss; this is important, especially to those users that are new to the dieting plan and needs to understand more about how dieting works. The app should explain the scientific and technical studies for users to understand how the app manages diet according to studies. Like the previous system, the app should also include features that are interesting that can create continuous user interest using the app, such as a timelapse of pictures of themselves and collecting streak points to keep the streak throughout their dieting progress.

2.2.3 MyNetDiary

The app did not show the estimated days left for the user to achieve the targeted weight goal. Similarly to the previous method, the app does not explain how calorie counting works in weight loss management; this is significant, especially for users who are new to the dieting plan and need to learn more about how dieting works. Users should be able to grasp how the app regulates nutrition based on scientific and technological research if the app explains the scientific and technical studies. The app should have engaging elements that will keep users interested in using the app, such as a timelapse of themselves and accumulating streak points to keep the streak going throughout their dieting process.

CHAPTER 3 SYSTEM METHODOLOGY/APPROACH

This chapter describes the system methodology used in the development of the Diet Manager app. The planning, analysis, design, implementation, and testing phases of the app's development are covered in this chapter using the System Development Life Cycle (SDLC) methodology. In addition, the topic of phased development is covered, emphasising how the app is split into two versions with different functionalities. The details of each phase are covered in-depth in each section of this chapter, giving a thorough understanding of the process used to create the Diet Manager app.

3.1 System Methodology

The methodology used to develop the proposed system will be based on the System Development Life Cycle (SDLC), which uses a phased development model. The goal of the SDLC is to provide an organised method for developing software, ensuring that the finished product is of high quality, satisfies user requirements, and is delivered within the allotted budget and time limitations. Its ability to reduce risks and support efficient resource management underpins its significance. The system's success and functionality are ensured by the five phases of the development process: planning, analysis, design, implementation, and testing. Each of these phases is essential to the systems' success.

3.1.1 Planning Phase

The planning phase will be gathering information on the existing system by reviewing other diet manager apps. This phase will determine all the basic functions and the expected output of the diet managing app, such as calorie counting, tracking diet progress functions etc. The project's goals, criteria, and specifications are set during this phase, along with its needs. Finding out about current methods and assessing their advantages and disadvantages can assist in determining the gaps that the new diet management software can address. To make sure the app satisfies the demands of those who are its intended audience, it is crucial to identify the core capabilities and desired results. In this phase, the project's feasibility is also evaluated, along with the cost, availability of resources, and timeline needed for success. Therefore, careful planning is essential to guarantee the project's success and to reduce the possibility of costly mistakes and delays during later stages of development.

3.1.2 Design Phase

The design phases will start with creating the diet manager app design of all the defined functions of the diet manager app. In this stage, it is required to visually translate the user interface and functionality of the app using the project's main objectives and sub-objectives. Then the project can begin by developing a thorough design for each of the specified features of the diet manager app, including the diet management module, the weight tracking tool, the summary report system, the reminder tool, the timelapse tool, and the streak system. The architecture, layout, navigation, and user experience of the app will all be included in the design. This step is crucial because it makes sure that the app's features and functionalities work together effectively and efficiently, resulting in a user-friendly and simple to use app.

3.1.3 Implementation Phase

The implementation phases will start implementing the diet manager functions according to the design phase. All the functions of the app will be implemented at this phase. In this phase, the project will place the Diet Manager's features and functions into operation in accordance with the design guidelines established in the previous phase. In order to produce an app that is fully functioning, this phase entails writing the app's code and integrating all of its features and components. The implementation stage is crucial since it is here that the software takes shape and where its original idea materialises. The project will ensure that all app's features perform well together and are user-friendly. The Diet Manager app were developed and improved during the implementation phase to make sure it satisfied the project's goals and user requirements. The Diet Manager app were checked and ensured that the app runs properly on the Android platform during this phase. The Diet Manager app will be prepared for testing and deployment after the implementation step has been successfully completed. The implementation stage is a crucial step in the creation of the Diet Manager app, and its accomplishment defines the functionalities that satisfies the objectives of the project.

3.1.4 Testing Phase

The testing phase will run and test the diet manager app to see if all the functions work. The Diet Manager app's testing phase is an important stage in its development. The project team will use the app during this phase to test and confirm that all of the features and functions operate as planned. Finding any mistakes, flaws, or glitches that might have appeared during the development and implementation phases is the goal of the testing phase. The system will be tested using integrated testing, which verifies that all of the app's elements and interactions operate as intended. This kind of testing is necessary to make sure the app performs as intended and achieves the project's goals. The testing stage is crucial since it makes sure the Diet Manager app were prepared for deployment. Testing guarantees that the app is dependable, safe, and operates well on the Android platform.

3.1.5 Phased Development

The Diet Manager app's Phased Development is a crucial strategy that manage the project's complexity. The development of the overall system will break into two series of versions that will be developed in series. The first version of the system consisted of all the basic functions of the diet app, such as calorie counting calculation, recording calorie intake daily, tracking diet progress with the user's weight and reminders for the user to input food intake in every meal or end of the day. Prior to moving to the second version, the app will be tested in the first version to make sure it satisfies the project's objectives and user needs. The second version will implement advanced features such as providing diet teachings and guidance for the user to follow, an explanation of how dieting works to the user and a display summary report for the user to view overall performance.

3.2 System Design Diagram

3.2.1 Use Case Diagram

The following (Refer to Figure 3.1) use case diagram described all the functions that the user can perform on the Diet Manager app.



Figure 3. 1 Use Case Diagram

••••

This use case in Figure 3.1 described all the available functionality implemented in the Diet Manager app. The actor in the use case is a any user who are dieting that uses the Diet Manager system an an android device. The user can perform all six of the Diet Manager's main functionalities in the android device which are calculate calories intake, record diet progress, read summary report, reminf user input calories intake, create timelapse and collect streak points.

3.2.2 Activity Diagram

The following activity diagram (Refer Figure 3.2) describe the processes of action carried out by each use case function.

i. Calculate calories intake



Calculate calories intake

Figure 3. 2 Calculate the calories intake function

The Figure 3.2 described the calculate calories inateke function of the Diet Manager. The process begin by user entering their daily calories intake into the app, Then inputted user calories will be read by the app. The app will deduct the daily calories intake limit with the inputted user calories to produce and display the remaining calories. Add the end of the day, app will create a new calories record from the user added calories.

ii. Record diet progress



Visual Paradigm Online Free Edition

Figure 3. 3 Record diet progress function

The Figure 3.3 describe the record diet progress function. The process begin by user input the user weight into the app. The app then read the user weight date together with the user added weight to create a new weight record. Then the app will add and create a new weight record to the app.

iii. Read summary report



Read summary report

Figure 3. 4 Read summary report function

CHAPTER 3

The Figure 3.4 describe the read summary report function. The app will read all the user added weight record then with it to create estimated duration for user to reach weight goal. Then the app will create a weight graph and insert all the user added weight and also the estimated estimated date for user to targeted weight goal into the line graph. The line graph are displayed to the user. A short summary report will explain about the graph and the target weight goal date of the user.

iv. Remind the user to input calorie intake

Remind user input calorie intake



Figure 3. 5 Remind user input calorie intake function

The Figure 3.5 describe the remind user input calories intake function. The process begin the app checking if the user have inputted the daily food intake for the day. If the user did not, then the app will create a reminder notifuication to remind user, if user did, then nothing will happen.

v. Create timelapse



Create timelapse

Figure 3. 6 Create a timelapse function

The Figure 3.6 describe the create a timelpase function. The process begin by user input use picture into the app. Then will all the user inputed picture, to create and add a timelpase record. Then the app will create the timelapse video. The timelapse can be displayed.

vi. Collect streak points

Collect streak points



Figure 3. 7 Collect streak function

The Figure 3.7 describe the collect streak function. The process begin by the app check for calories intake limit, if the calories intake is within the limit, then user are entitled to redeem one streak point for the day, and the user's streak point record were updated

and displayed. If the user exceeded calories intake limit, then the app will deduct 5 streak points as penalties.

3.3 Project Time

The duration required for the entire development of the diet app manager will be divided into two semesters, with each version of the app for each semester to develop. The approximate duration of the first version of the app will be completed within six weeks (Refer to Figure 3.8), while the second version will require 12 weeks to complete (Refer to Figures 3.9 & 3.10). The expected deliverables for the first semester will be the prototype of the app displaying the UI and the page navigation for all the functions in the app, as well as three prototype diet functions to be implemented into the diet app. The functions include the streak function, timelapse function, and calorie counting function. The expected deliverables for the next semester will be all six full working functions of the diet app with UI design implemented.





Figure 3. 8 Gantt Chart for the First Version of the app in the first semester









CHAPTER 4 SYSTEM DESIGN

This chapter presents the system design of the Diet Manager application, focusing on its database structure, sitemap, user interface (UI) design, and client-server architectural design. This chapter aims to provide a comprehensive understanding of the application's structure and functionalities by discussing the Firebase Firestore and Firebase Realtime Database structures in tree-like formats, outlining the sitemap and main pages, delving into the UI design of key screens, and explaining the client-server architectural design. By discussing these aspects, this chapter intends to provide a comprehensive overview of the application's structure and its underlying functionalities.

4.1 Firebase Firestore Tree-Like Structure

The database used in the project is Firebase Firestore and Firebase Realtime Database. The Firebase Firestore and Firebase Realtime Database structures were represented using Firebase Tree-Like Structure.

The diagram below shows the Firebase Firestore database structure in Tree-Like Structure.



Figure 4. 1 Firebase Firestore Tree-Like

Cloud Firestore is a document-oriented, NoSQL database. There are no tables or rows like there are in a SQL database. Instead, data is saved as documents that are arranged into collections. Each document has a collection of key-value pairs. All documents are kept in collections. Documents can include subcollections and nested objects, both of which can contain simple fields like strings or complex objects like lists. In the Figure

above (Figure 4.1), the dark blue box was used to represent the Collection, and the light blue box was used to represent the Documents.

The "UserAccount" is a Collection, and within it consists of a Document called "usersid". And within the "users-id" Document consist of the field "emails".

The "users" is a Collection, and within it consists of a Document called "users-id". And within the "users-id" Document consist of the fields "age", "amountOfDays", "currentStreak", "dailyCaloriesIntakeLimit", "gender", "hasRedeemedStreakToday", "lastRecordDate", "lastRedeemedDate", "name", "speed", "startingDietDate", "targetWeight", "totalCaloriesConsumed" and "weight". Then the "pictures" Collection is nested within the "users-id" Document and has the "pictures-id" Document that contains the fields "date" and "imageUrl".

4.2 Firebase Realtime Database Tree-Like Structure

The figure below shows the Firebase Realtime Database structure in Tree-Like Structure.



Figure 4. 2 Firebase Firestore Tree-Like

Firebase Realtime Database is a cloud NoSQL database. Data is kept in the Firebase Realtime Database as JSON objects. Each node in the tree, which is kept in a hierarchical tree-like structure, corresponds to a JSON item. Each node has the ability to have one or more child nodes, each of which can have more children of its own. Keyvalue pairs representing the data are kept in Firebase Realtime Database. Each value can be any data type that is JSON-compatible, including a text, integer, Boolean, object, or array. Each key is a string that uniquely identifies a node in the tree.

The "users" is a Parent Node, and within it consists of a Child Node called "users-id". And within the "users-id", Child Node consist of the Parent Node "food" and "weights" of the Child Node "food-id" and "weight-id". The "food-id" node contained the fields "calories", "date", and "food". And the "weight-id" node contained the fields "datetime" and "weights".

4.3 Sitemap of Diet Manager

The diagram below shows the sitemap of the Diet Manager application.



Figure 4. 3 Sitemap of Diet Manager

Figure 4.3 above presents the sitemaps of the Diet Manager Application. It provides an overview of the Diet Manager application layout, including the element hierarchy and the link between content and pages. Figure 4.3 above presents the sitemaps of the Diet Manager Application. The application contains some of the main pages, which consist of the application's main functions such as "Record Page", "Progress Page", "Timelapse Page", and "Streak Page" under the "Bottom Navigation", as well as another one main function which is "Reminder Page", and "Welcome Page" under the "Left Navigation Drawer".

Below is the breakdown of how each page fulfilled the project sub-objectives; please refer to Chapter 1 1.2 Project Sub-Objectives:

• The "Record Page" consist of the functionality of the first project sub-objective. It also can navigate to view past consumed food in the "Record Food History Page" and further can view each food item in detailed in the "Food Detail Page".

- The "Progress Page" consist of the functionality of the second and third project sub-objective.
- The "Reminder Page" consists of the functionality of the fourth project subobjective.
- The "Timelapse Page" consists of the functionality of the fifth project subobjective.
- The "Streak Page" consists of the functionality of the fourth project subobjective.

Other pages, such as the "Sign-in Page", "Sign-up Page", and "Forgot Password Page", are the application user authentication pages, which let users sign up for an account and sign in to their account in the app as well as enable the user to reset their password if in an event where the user forgets their account password. The user is also able to sign out by going to the "Sign Out" under the "Left Navigation Drawer" of the app. As well as the "Welcome Page" provides new users with the main key features of the Diet Manager upon signing up for an account, as well as provides an explanation of how dieting calculation works by including the Harris-Benedict equation used for weight loss.

4.4 System UI Design

This sub-section discussed the UI design for the screens of each of the main functions of the Diet Manager.

4.4.2 UI Design of Record Page screen

The following screen is the UI Design of the Record Page.

Record You	ır Weight	
roday's Remaining Co	alories	
689 ko	cal	
Total Calories Consur	med: 603	kcal
603 / 129	92 kcal	
	View Calorie	s History
Apples	52 k	cal
Bread White Wheat	119 k	cal
Lowfat Chocolate Milk	152 k	cal
Sausage Pizza	280 k	cal
A Record		6

Figure 4. 4 UI Design of Record Page

The Progress Page screen contained these elements:

- 1. Target Weight, which is the user-specified target weight goal
- 2. Weight Goal Date, which is the estimated date the user will achieve the diet goal

- 3. The Weight Graph is a graph that plots all the weight records with dates added by the user and the estimated weight goal date at most further dates.
- 4. Weight Record List, which is a list of all user-recorded weight lists.
- 5. Add Weight Button, which lets the user add a new weight record.
- 6. Progress Bar to track the amount of weight loss against the total weight to lose.

The following screen is the UI Design for the Food History List page and Food Details Page.





Figure 4. 6 UI Design of Food History Details Page from Food History List Page

The View Calories History navigates to the Food History List, where it lists all the past consumed calories item arranged by the dates.

Selecting any of the consumed calorie items will display a breakdown of all the foods consumed and their calories for the selected day.

4.4.3 UI Design of Progress Page screen

The following screen is the UI Design for the Progress Page page.

	Progre	SS	
Target W	eight		
50	.0 k	g	
Weight G	oal Date:	09/05/	2023
	2.0 0 / 5.0	0 kg	
65			
60 -			
55 55	53 53	Weigh	t Goal
54.5	53.4		y 9 : 50 D
45			
40	Apr 1	21	
28/03/2	023	53.0	kg
24/03/2	023	53.4	+
A	* Progress		6

Figure 4. 7 UI Design of Progress Page

The Progress Page consists of these elements:

- 1. Target Weight, which is the user-specified target weight goal
- 2. Weight Goal Date, which is the estimated date the user will achieve the diet goal
- 3. The Weight Graph is a graph that plots all the weight records with dates added by the user and the estimated weight goal date at most further dates.
- 4. Weight Record List, which is a list of all user-recorded weight lists.
- 5. Add Weight Button, which lets the user add a new weight record.
- 6. Progress Bar to track the amount of weight lost against the total weight to lose.

4.4.4 UI Design of Reminder Page screen

The following screen is the UI Design for the Reminder Page page.



Figure 4. 8 UI Design of Reminder Page

The Reminder Page consists of these elements:

- 1. Edit Time button, for the user set a time for daily reminders.
- 2. Notification On toggle, to enable of disable

4.4.5 UI Design of Timelapse Page screen

The following screen is the UI Design for the Timelapse Page page.



Figure 4. 9 UI Design of Timelapse Page

The Timelapse Page consists of these elements:

- 1. The Thumbnail Photos Gallery, which displays all the photos taken by the user reverse chronological order in a grid view.
- 2. The Take Photo button, which lets user takes a photo and add to the gallery.
- 3. The Select All Photo button, which lets user to select all the photos in the gallery to create the Timelapse video.
- 4. The Download Pictures button, which only appears when pictures are selected and lets user to download the selected pictures.
- 5. The Create Timelapse Video button, which only appears when pictures are selected and lets user to create timelapse video using all the selected photos.
- 6. The Cancel Selections button, which cancels the photo selection mode.

4.4.6 UI Design of Streak Page screen

The following screen is the UI Design for the Streak Page page.



Figure 4. 10 UI Design of Streak Page

The Streak Page consists of these elements:

- 1. Current Streak count, which displays the total cumulative streak count the user has collected using the app daily.
- 2. The Highest Streak Score, which displays the highest streak cumulative streak count the user has ever achieved.
- 3. The Collect Streak icon button, which limits the user to collecting only one streak point daily.

4.4.7 UI Design of User Profile Page screen

The following screen is the UI Design for the User Profile Page page.

User Details		
Hi, I'm Chee Wah, 25 years		
old, male, and my current		
weight is 55.0 kg . I aim to		
achieve 50.0 kg within 70		
days by limiting my calories		
intake to 1268 kcal daily. My		
diet journey begins on		
19/4/2023, and		
this is my story!		
Name		
Chee Wah		
Male		
Age		
Current Weight		
55.0		
Target Weight Goal		
50.0		
Physical Activity Level		
Starting Diet Date		
19/4/2023		

Figure 4. 11 UI Design of User Profile Page

CHAPTER 4

The User Profile Page consists of these elements:

- 1. User Diet Goal dialog, a simple SMART diet goal created for user to get an overview of their dieting journey.
- 2. Name, for user to enter their name
- 3. Gender, for user to select gender
- 4. Age, for user to enter their age
- 5. Current Weight, for user enter their current weight
- 6. Target Weight Goal, for user to enter their targeted weight goal
- 7. Physical Activity Level, for user to select their Physical Activity Level
- 8. Starting Diet Date, for user to select the starting diet day
- 9. Save Button, to save all user details.

4.5 Client Server Architectural Design



Figure 4. 12 Client Server Architectural Design of the Diet Manager application

The Client, which is the Android Device running the Diet Manager application, were responsible for making requests for data from the Firebase server through the Internet, processing or otherwise modifying the requested data and updating the Android Device's user interface (UI) as necessary. The Client retrieves data from the Firebase server and processes the data within the Client, which is the Diet Manager functions.

The Diet Manager's functions that send and receive the data for processing from the Firebase are Calculate Calories Function, Record Weight Function, Diet Summary Function, Reminder Function, Streak Function and Timelapse Function. (Refer to Chapter 5.4 for details of each of the Diet Manager's functions)

And Firebase sends and receives the data from the Diet Manager's functions to its cloud services which are Firebase Realtime Database, Firestore Authentication, Firebase Storage, and Firebase Firestore.

The Firestore Authentication was mainly used for authenticating the user before the user accessed the application. The user authentication process takes place within the

CHAPTER 4

Firebase Authentication service. (Refer to Chapter 5.4.1 for details of the User Authentication of the Diet Manager)

The Firebase Cloud Storage was mainly used for storing and receiving media files for the Timelapse Function within the Diet Manager application. (Refer to Chapter 5.4.7 for details of the Timelapse function)

The Firebase Realtime Database and Firebase Firestore were mainly used for storing user data for the Calculate Calories Function, Record Weight Function, Diet Summary Function, Streak Function and Timelapse Function of the Diet Manager. (Refer to Chapter 5.4 for details of the Diet Manager's functions)

CHAPTER 5 SYSTEM IMPLEMENTATION

5.1 Hardware Setup

The hardware that will be used in this project is a computer and mobile device running Android OS. The computer is required to run Android Studio, which is an android development environment for the Android operating system to create the mobile application for the Diet Manager android application. The android mobile device is required to simulate the real-life experience of users when using the mobile application on a physical device and to utilize the functions that are only available on the phone, such as the touch screen.

The android mobile device that was used were connected to the computer using a USB Type-C cable throughout the development and testing of the Diet Manager app.

Description	Specifications
Model	Acer Nitro AN515-52
Processor	Intel Core i5-8300H
Operating System	Windows 11
Graphic	NVIDIA GeForce GTX 1050 2GB GDDR5
	Intel UHD Graphics 630
Memory	8GB DDR4 RAM
Storage	1 TB M.2 NVMe SSD

Table 5.1	Specifications	of Laptop
-----------	----------------	-----------

Description	Specifications
Model	Samsung Galaxy Note 10 Lite
Processor	Exynos 9810 SoC
Operating System	Android 12
Graphic	Mali-G72 MP18
Memory	8GB RAM
Storage	128GB

Table 5.2 Specifications of Mobile Phone

5.2 Software Setup

This section discussed all the software required for the implementation of the Diet Manager project.

5.2.1 Flutter Set-up

The Flutter SDK were required to be installed on the Windows computer. The Flutter SDK files were downloaded from Flutter's official SDK download page [4]. You may refer to Chapter 5.3.1 for further details on the settings and configurations required for Flutter to be set-up for running the project.

5.2.2 Java Set-up

The Java JDK 11 software is required to be installed on the computer. The Java JDK 11 can be downloaded from Java's official download website [5].

You may refer to Chapter 5.3.2 for further details on settings and configurations required for Java to be set-up for running the project.

5.2.3 Android Studio Set-up

The Android Studio will be the IDE used for the development of the Diet Manager android application. The Android Studio can be downloaded from the Android Studio's official download page [4]. The Android Studio version used for the project is Android Studio Electric Eel 2022.1 Canary 10.

You may refer to Chapter 5.3.3 for further details on settings and configurations required for Android Studio to be set-up for running the project.

5.2.4 Firebase Set-up

Firebase is a backend cloud computing service that will be used for the Diet Manager app databases, authentications, and storage. The Firebase services can be accessed on the Firebase's official website, a Firebase account is required.

You may refer to Chapter 5.3.4 for further details on the settings and configurations required for Firebase to be set-up for running the project.
5.2.5 Mobile Device Set-up

Connect the android mobile device to the computer using a USB cable, ensuring that the computer can connect and detect the connect mobile devices. Certain mobile devices are required to install the Android USB Driver for the connected mobile device to function properly. In this case, the project mobile device used is a Samsung Note 10 Lite Android device, which is required to install the Samsung Android USB Driver for the Samsung Developers website [7].

You may refer to Chapter 5.3.5 for further details on settings and configurations required for mobile devices to be set-up for running the project.

5.3 Setting and Configurations

This section describes each of the software settings and configurations required for all the software mentioned earlier in Chapter 5.2.

5.3.1 Flutter Set-up

Extract the contents of the downloaded Flutter SDK zip file to a new folder named src, and place it in the root folder of Local Disk C. The Flutter SDK files directory should be C:\src\flutter.

Add the Flutter SDK file to the PATH environment variable by following these steps:

Click on the Start menu and search for "environment variables."



Figure 5. 1 Search "environment variables" in the start menu

The System Properties window will appear; on the Advanced tab, click on the "Environment Variables" button

System Propertie	25				×
Computer Name	Hardware	Advanced	System Protection	Remote	
You must be lo	gged on as a	an Administrat	or to make most of	these changes.	
Visual effects,	, processor s	cheduling, m	emory usage, and v	irtual memory	
			(<u>S</u> ettings	
User Profiles					
Desktop settir	ngs related to	o your sign-in			
			(Settings	
Startup and R	ecovery				
System startup	p, system fail	lure, and deb	ugging information		
			(Settings	
			Environm	ent Variables	
		ОК	Cancel	Apply	

Figure 5. 2 System Properties Window

The Environmental Variables Window will appear. Under "User Variables," find the "Path" variable and click on "Edit."

Variable	Value
JAVA HOME	C:\Program Files\Java\jdk-11.0.17
OneDrive	C:\Users\leech\OneDrive - Universiti Tunku Abdul Rahman
OneDriveCommercial	C:\Users\leech\OneDrive - Universiti Tunku Abdul Rahman
Path	C:\Users\leech\AppData\Local\Microsoft\WindowsApps;C:\Progra
TEMP	C:\Users\leech\AppData\Local\Temp
ТМР	C:\Users\leech\AppData\Local\Temp
	New Edit Delete
stem variables	New Edit Delete
stem variables Variable	New Edit Delete
stem variables Variable ComSpec	New Edit Delete Value C:\Windows\system32\cmd.exe
stem variables Variable ComSpec DriverData	New Edit Delete Value C:\Windows\system32\cmd.exe C:\Windows\System32\Driver>DriverData
stem variables Variable ComSpec DriverData JAVA_HOME	New Edit Delete Value C:\Windows\system32\cmd.exe C:\Windows\System32\Drivers\DriverData C:\Program Files\Java\jdk-11.0.17\
stem variables Variable ComSpec DriverData JAVA_HOME NUMBER_OF_PROCESSORS	New Edit Delete Value C:\Windows\system32\crnd.exe C:\Windows\System32\Drivers\DriverData C:\Program Files\Java\jdk-11.0.17\ 8
stem variables Variable ComSpec DriverData JAVA_HOME NUMBER_OF_PROCESSORS OS	New Edit Delete Value C:\Windows\system32\crud.exe C:\Windows\System32\Driver\DriverData C:\Program Files\Java\jdk-11.0.17\ 8 Windows_NT
stem variables Variable ComSpec DriverData JAVA_HOME NUMBER_OF_PROCESSORS OS Path	New Edit Delete Value C:\Windows\system32\cmd.exe C:\Windows\System32\Drivers\DriverData C:\Program Files\Java\jdk-11.0.17\ 8 Windows_NT C:\Program Files\Common Files\Oracle\Java\javapath;C:\Program
stem variables Variable ComSpec DriverData JAVA_HOME NUMBER_OF_PROCESSORS OS Path PATHEXT	New Edit Delete Value C:\Windows\system32\cmd.exe C:\Windows\System32\Drivers\DriverData C:\Program Files\Java\jdk-11.0.17\ 8 Windows_NT C:\Program Files\Common Files\Oracle\Java\javapath;C:\Program .COM:.EXE:.BAT.:CMD:.VBS:.VBE:.JS:.JSE:.WSF:.WSFIMSC

Figure 5. 3 Environmental Variables Window

The Edit environment variable Window will appear. Click on "New" and add the path to the Flutter bin folder, which is located at C:\src\flutter\bin.

Edit environment variable	×
%USERPROFILE%\AppData\Local\Microsoft\WindowsApps	New
C:\Program Files\Intel\WiFi\bin\	
C:\Program Files\Common Files\Intel\WirelessCommon\	Edit
C:\src\flutter\bin	
C:\Program Files\Java\jdk-11.0.17\bin	Browse
	Delete
	Move Up
	Move Down
	Edit text
]
ОК	Cancel

Figure 5. 4 Edit environment variable Window

5.3.2 Java Set-up

The environmental variables are required to be set up following these steps: Click on the Start menu and search for "environment variables."



Figure 5. 5 Search "environment variable" in the start menu

The System Properties Window will appear. On the Advanced tab, click on the "Environment Variables" button.

System Properties				\times
Computer Name Hardware	Advanced	System Protection	Remote	
You must be logged on as a Performance Visual effects, processor so	n Administrai cheduling, m	tor to make most of ti emory usage, and vir	hese changes. tual memory Settings	
User Profiles Desktop settings related to	your sign-in		S <u>e</u> ttings	
Startup and Recovery System startup, system failu	ure, and deb	ugging information	Se <u>t</u> tings	
		Environme	ent Variables	
	ОК	Cancel	Apply	

Figure 5. 6 System Properties Window

The Environment Variables Window will appear. Under "User Variables," on "New...",

Variable	Value			
JAVA_HOME	C:\Program Files\Java\jdk-11.0.17			
OneDrive	C:\Users\leech\OneDrive - Universiti Tunku Abdul Rahman			
OneDriveCommercial	C:\Users\leech\OneDrive - Universiti Tunku Abdul Rahman			
Path	$C: \label{eq:constraint} C: \label{eq:constraint} C: \label{eq:constraint} Users \label{eq:constraint} C: eq:con$			
TEMP	C:\Users\leech\AppData\Local\Temp			
TMP	C:\Users\leech\AppData\Local\Temp			
	New Edit Delete			
the second states	New Edit Delete			
stem variables	New Edit Delete			
stem variables Variable	New Edit Delete			
stem variables Variable ComSpec	New Edit Delete Value C:\Windows\system32\cmd.exe			
stem variables Variable ComSpec DriverData	New Edit Delete Value C:\Windows\system32\cmd.exe C:\Windows\System32\Drivers\DriverData C:\Windows\System32\Drivers\DriverData			
stem variables Variable ComSpec DriverData JAVA_HOME	New Edit Delete Value C:\Windows\system32\cmd.exe C:\Windows\System32\Drivers\DriverData C:\Program Files\Java\jdk-11.0.17\			
stem variables Variable ComSpec DriverData JAVA_HOME NUMBER_OF_PROCESSORS OS	New Edit Delete Value C:\Windows\system32\cmd.exe C:\Windows\System32\Drivers\DriverData C:\Program Files\Java\jdk-11.0.17\ 8			
stem variables Variable ComSpec DriverData JAVA_HOME NUMBER_OF_PROCESSORS OS Deth	New Edit Delete Value C:\Windows\system32\cmd.exe C:\Windows\System32\Drivers\DriverData C:\Program Files\Java\jdk-11.0.17\ 8 Windows_NT C:\Program Files\Common Files\Oracle\Java\jjuvanathiC\Program			
stem variables Variable ComSpec DriverData JAVA_HOME NUMBER_OF_PROCESSORS OS Path DATHEXT	New Edit Delete Value C:\Windows\system32\cmd.exe C:\Windows\System32\Drivers\DriverData C:\Program Files\Java\jdk-11.0.17\ 8 Windows_NT C:\Program Files\Common Files\Oracle\Java\javapath;C:\Program COM: EXE: BAT: CMD: VBS: VBE: IS: ISE: WSE: WSE: MSE:			
stem variables Variable ComSpec DriverData JAVA_HOME NUMBER_OF_PROCESSORS OS Path PATHEXT	New Edit Delete Value C:\Windows\system32\cmd.exe C:\Windows\System32\Drivers\DriverData C:\Program Files\Java\jdk-11.0.17\ 8 Windows_NT C:\Program Files\Common Files\Oracle\Java\javapath;C:\Program .COM:.EXE:.BATCMD:.VBS:.VBE:.JS.JSE:.WSF:.WSH:.MSC			

Figure 5. 7 Environment Variables Window

The Edit User Variable Window will appear. Add the new variables as follows, and click "OK" to save the environmental variable:

Variable name: JAVA_HOME

Variable value: C:\Program Files\Java\jdk-11.0.17

Edit User Variable		×
Variable name:	JAVA_HOME	
Variable value:	C:\Program Files\Java\jdk-11.0.17	
Browse Directory	Browse File	OK Cancel

Figure 5. 8 Edit User Variable Window

5.3.3 Android Studio Set-up

Implementing Flutter SDK into Android Studio

Run the Android Studio, and a welcome page will appear, navigate to the Plugins tab on the left side of the window, and under the Marketplace tab, search for "Flutter" and click on install. This will install the Flutter plugins in Android Studio.

🛛 🐱 🛛 Welcome to Android Studio					
		Marketplace I	nstalled 🌣		
Electric Eel 2022.1.1 Patch 2	Q+ Flutter		/ 5	uttor	
Projects Customize	Search Results (78)	Sort By: Relevance		13.7M 🕁 3.90 flutter.de Programming Language	
Plugins Learn Android Studio	Fj FlutterJsonBeanf ±311.1K ☆4.86	Fact Install	Plugin homepa Support for der developers an deploy cross-p for both Andro	age > veloping Flutter applicatio easy and productive way latform, high-performan id and iOS. Installing this	ons. Flutter gives to build and ce mobile apps plugin will also
	 ▲ 179.3K ☆ 4.48 FlutterAssetsGer ▲ 56K ☆ 4.67 	nera	For some tools JxBrowser to di complies with	plugin. , this plugin uses Chromi splay content from the w LGPL and offers an option	um through reb. JxBrowser n to replace
	Flutter Intl ≟ 437.3K ☆ 4.67	Install	• Find the J directory, • Use the in	handther component, re hxBrowser files stored in t , under /flutter-intellij/jxE hstructions and build scrij	he plugins prowser. pt from
			JxBrowse	r to relink it with modifie	d components.
*	flutter_json_form <u>↓</u> 76.9K	nat Install	Size: 13.32 MB		

Figure 5. 9 Search Flutter in the Plugins tab of Welcome to Android Studio Screen

Do the same for the "Dart".

CHAPTER 5



Figure 5. 10 Search Dart in the Plugins tab of Welcome to Android Studio Screen

Next, go to the command prompt. And type in "flutter doctor – android-licenses." to complete the Flutter SDK installation.



Figure 5. 11 Type "flutter doctor - android-licenses." in Command Prompt Screen

On the command prompt again, type in "flutter doctor" and ensure the it displayed ticked on everything to show that flutter had been installed successfully set up on the computer.

⊡ C:\Windows\system32\cmd.e: × + ∨			×
Microsoft Windows [Version 10.0.22621.1555] (c) Microsoft Corporation. All rights reserved.			
<pre>C:\Users\leech>flutter doctor Doctor summary (to see all details, run flutter doctor -v): [/] Flutter (Channel stable, 3.7.11, on Microsoft Windows [Vers 10.0.22621.1555], locale en-MY) [/] Windows Version (Installed version of Windows is version 16 [/] Android toolchain - develop for Android devices (Android SI 33.0.2) [/] Chrome - develop for the web [X] Visual Studio - develop for Windows X Visual Studio not installed; this is necessary for Window development. Download at https://visualstudio.microsoft.com/downloads/ Please install the "Desktop development with C++" workloa including all of its default components [/] Android Studio (version 2022.1) [/] Connected device (3 available) [/] HTTP Host Availability</pre>	sion 9 or 1 DK ver ws /. ad,	higher rsion	.)
C:\Users\leech>			

Figure 5. 12 Type "flutter doctor" in Command Prompt Screen

Then go to Android Studio, File - > Settings -> Appearance & Behavior -> System Settings -> Android SDK -> SDK Tools, and ensure that the Android SDK Commandline Tools is installed.

🞽 Settings		
Q+	Appearance & Behavior >> System Settings >> Android SDK	\leftarrow \rightarrow
✓ Appearance & Behavior	Manager for the Android SDK and Tools used by the IDE	
Appearance	Android SDK Location: C:\Users\leech\AppData\Local\Android\Sdk	
Menus and Toolbars V System Settings	SDK Platforms SDK Tools SDK Update Sites	
HTTP Proxy	Below are the available SDK developer tools. Once installed, the IDE will automatically c "show package details" to display available versions of an SDK Tool.	neck for updates. Check
Date Formatr		
Undeter	Android SDK Build-Tools 34-rc3	Installed
Updates	NDK (Side by side)	Not Installed
Process Elevation	Android SDK Command-line Tools (latest)	
Passwords	CMake	Not Installed
Android SDK	Android Auto API Simulators 1	Not installed
Memory Settings	Android Auto Desktop Head Unit Emulator 2.0	Not installed
File Colors	Android Emulator 33.1.2	Installed
Scones	Android Emulator Hypervisor Driver (installer) 2.0.0	Not installed
Nutricia	Android SDK Platform-Tools 34.0.1	Installed
Notifications	Google Play APK Expansion library 1	Not installed
Quick Lists	Google Play Instant Development SDK 1.9.0	Not installed
Path Variables	Google Play Licensing Library	Not installed
Keymap	Google Play services 49	Installed
> Editor	Google Web Driver 2	Not installed
Plugins	■ V Intel x86 Emulator Accelerator (HAXM installer) 7.6.5	installed
> Version Control	Lavout Inspector image server for API 29-30 6	Not installed
Ruild Execution Deployment	Layout Inspector image server for API 31-34 2	Update Available: 3
	Layout Inspector image server for API S 3	Not installed
 Languages & Frameworks 		
> Tools	- V Hide	Disolete Packages 🔄 Show Package Details
Advanced Settings		
?		OK Cancel Apply
🖕 sogsas 🛛 y ripp quanty morgina	estitus grippinspectari a transmi e antrimitari	

Figure 5. 13 SDK Tools Screen in Android Studio Settings window

Importing the provided Diet Manager project files into the Android Studio Extract the provided project files zip file.

On the Android Studio, select "Open" and browse to the extracted project folder. The project folder should be "crudfirestore".



Figure 5. 14 Open files in the Project screen in Android Studio

5.3.4 Firebase Set-up

At the Firebase website, add Project and name the project to any name, e.g. crudfirestore

×	Create a project (Step 1 of 3)
	Let's start with a name for
	your project [®]
	Project name
	cruamrestore
	Crudfirestore-86d35
	Continue

Figure 5. 15 Add Project screen on the Firebase website

Disable Google Analytics, as it will not be required for this project, and click on the "Create Project" button.

(Create a project (Step 2 of 2) Google Analytics						
(
1	for yo	ur Firebase	pro	je	ct		
0 r 0	Google Analyt eporting, and Config, A/B Te	ics is a free and unlimited more in Firebase Crashlyt ssting, and Cloud Function:	analytics ics, Cloud s.	solu Mes	tion that enables targeting, saging, In-App Messaging, Remote		
G	oogle Analytics	enables:					
:	× A/B testin	g (2)		\times	Crash free users ⑦		
:	X User segn	nentation & targeting across	0	×	Event based Cloud Functions triggers ③		
	Firebase (roducts		×	Free unlimited reporting ⑦		
¢	Enable Recomm	Google Analytics for this p ended	project				
F	Previous				Create project		

Figure 5. 16 Add Project Screen 2 to the Firebase website

On the left side of the tab, select "All Products" to navigate to the Firebase products and features page.



Figure 5. 17 Firebase main page

Firebase Authentication set-up

On the Firebase products and features page, select Authentication to set up an Authentication service for the project.

Under the Sign-in method tab, select "Email/Password".

ers	Sign-in method	Templates	Usage	Settings	Extension	
s	Sign-in providers					
	Get started method	with Fire	base Autl	h by addi	ng your firs	t sign-in
	Native providers		Additional pro	viders	Custom pr	oviders
	Email/Pas	sword	G Google		(() ()	
	📞 Phone		Facebo	ok	🙆 sa	
	Anonymou	ıs	Þ Play Ga	mes		
			😗 Game C	enter		
			Apple			
			O GitHub			
			Microso	oft		

Figure 5. 18 Firebase Authentication page

Then enable Email/Password toggle, and click "Save" to enable Email/Password as a authentication method for the Diet Manager app.



Figure 5. 19 Firebase Authentication Sign-in providers page

CHAPTER 5

The green ticked status "Enabled" means it has been set up successfully.

n-in providers	
	Add new provide
Provider	Status
Email/Password	C Enabled

Figure 5. 20 Firebase Authentication Sign-in Provider Status

Firebase Firestore Set-up

The Firebase Firestore is one of the firebase database services that will be used for storing data for the Diet Manager app.

On the Firebase products and features page, select Cloud Firestore to set up Firestore service for the project.

A pop-up will appear; select "Start in Production Mode", and click "Next"

Create database		×
1 Secure rules for Cloud Firestore	2 Set Cloud Firestore location	
After you define your data structure, you wi l <u>Learn more</u> [2]	I need to write rules to secure your data.	
Start in production mode Your data is private by default. Client read/write access will only be granted as specified by your security rules.	<pre>rules_version = '2'; service cloud.firestore { match /databases/{database}/documents { match /document**} {</pre>	
Start in test mode Your data is open by default to enable quick setup. However, you must update your security rules within 30 days to enable long-term client read/write access.	<pre>} } All third party reads and writes will be denied</pre>	
Enabling Cloud Firestore will prevent you from usin	g Cloud Datastore with this project Cancel New	ct

Figure 5. 21 Cloud Firestore Create database page

Choose the location to anywhere. Ensure to take note of the chosen location. Then click "Enable" to start creating the database.



Figure 5. 22 Cloud Firestore Create database page 2

On the Cloud Firestore page, navigate to the "Indexes" tab, and click on "Add Index",

loud F	irestore		
ta Rules	Indexes Usage 🍪 Extensions 📭		
Composite	Single field		
Collection	ID Fields indexed 🕥	Query scope	Add index Status
pictures	group Ascending date Ascendingname Ascending	Collection	Enabled
nictures		Collection	Enabled

Figure 5. 23 Indexes Tab on the Cloud Firestore page

Add the index according to the details below, and click "Create Index":

Cre	ate a composite index		×
Cloud	Firestore uses composite indexes for comp es (ex: combining equality and range operato	ound queries not already supported by single field ors).	
Collec	tion ID		•
pict	ures		
Fields At lea	to index st two fields required*		
1	userld	Ascending -	
2	timestamp	Ascending -	
Ad	d field scopes Collection		
•	For queries within a specific collection path		
0	Collection group		
	For queries across all collections with the same	e collection ID	-
		Cancel Create in	dex
		Creation time depends on the amount of data being in	ndexed

Figure 5. 24 Composite Index Details page

Firebase Realtime Database Set-up

The Firebase Realtime Database is one of the databases used for the Diet Manager to store ser data in the app.

On the Firebase products and features page, select Realtime Database to set up Firebase Realtime database service for the project.

Select the Realtime Database location,

CHAPTER 5

Set up database	×
1 Database options — 2 Security rules	
Your location setting is where your Realtime Database data will be stored. Realtime Database location	
Singapore (asia-southeast1) -	
	Cancel Next
4	

Figure 5. 25 Firebase Realtime Database Set up database page

Select "Start in locked mode", then click "Enable".



Figure 5. 26 Firebase Realtime Database Set up database page 2

On the Realtime Database page, navigate to the "Rules" tab and edit and save the rules as follows:

Realtime Database								
Data	Rules Backups Usage 😻 Extensions 💌							
Edit rule	Monitor rules							
	Rules playground							
1								
2 •	{							
3 •	rules: { "users": {							
5 -	"\$uid": {							
6 •	"food": {							
7	".read": "auth != null && auth.uid == \$uid",							
8	".write": "auth != null && auth.uid == \$uid"							
10 -	/, "weights": {							
11	".read": "auth != null && auth.uid == \$uid",							
12	".write": "auth != null && auth.uid == \$uid"							
13	}							
14	}							
15	}							
17	}							

Figure 5. 27 Rules Tab on the Realtime Database page

Firebase Storage

Firebase Storage is one of the cloud storage used for the Diet Manager to store user data, such as media files in the app.

On the Firebase products and features page, select Cloud Storage to set up the Firebase Storage service for the project.

Select "Start in production mode and click on "Next".

Set up	Cloud Storage	2 Set Cloud Storage location
After you <u>Learn mo</u>	define your data structure, you w pre 🛛	ill need to write rules to secure your data.
St Yo rea gra rul	art in production mode ur data is private by default. Client ad/write access will only be anted as specified by your security es.	<pre>rules_version = '2'; service firebase.storage { match /b/{bucket}/o { match /{allPaths=**} { allow read, write: if false; } }</pre>
O St Yo en mu viti clii	art in test mode ur data is open by default to able quick setup. However, you ust update your security rules thin 30 days to enable long-term ent read/write access.	 All third party reads and writes will be denied
		Cancel Next

Figure 5. 28 Firebase Set up Cloud Storage page

The Cloud Storage location will be set automatically according to the previous location set in other Firebase services; click on "Done" to complete the set up.

Set up Cloud Storage	ge location	
Your location setting is where your default Cloud Storage bucket ar Cloud Storage location	d its data will be stored	d.
asia-southeast1		
This location was previously set, either during project creation or when setting up another service that supports location selection. Learn more	vation is not yet availa reased latency and bi <u>re to upgrade</u>	able. If your function and Storage Iling costs. <u>Learn more</u> [2] Cancel Done

Figure 5. 29 Firebase Set up Cloud Storage page 2

Implementing Firebase into the Android Studio project

This section should be the final step of setting up the Diet Manager project when all the required Firebase services for the project are set-up correctly and the provided Android Studio's project files containing the Diet Manager app has been imported into the Android Studio successfully. The Firebase services will be implemented into the Android Studio project to enable the Firebase services in the Diet Manager app. On the Firebase project's page, go to Project Overview on the left tab and select Project settings to Navigate to the Project Settings.



Figure 5. 30 Project settings in the Project Overview tab of the Firebase project page

Navigate to Your apps section under the General Tab, and select the Android icon,

to add an Android as the new platform for the Firebase project.

General Cloud Messaging	Integrations	Service accounts	Data privacy	Users and permissions
Vour project				
Project name	crudfirestoret	esting 🧪		
Project ID ⑦	crudfirestoret	esting-67776		
Project number ③	85873078645	5		
Default GCP resource location ③	asia-southeas	st1		
Web API Key	AlzaSyAzy1_8	8vkr5wdeu0Yw6-LMjf05	5-FkGgTr0	
Environment				
This setting customizes your projection	ot for different st	ages of the app lifecycle		
Environment type	Unspecified	1		
Public settings				
These settings control instances o	f your project sho	own to the public		
Public-facing name 🕥	project-85873	078645 🧪		
Support email	Not config	ured	•	
Your apps				
There are no apps in	your proje	ct		
Select a platform to get starte	1			
	_			
	05+ 🞽	(4)	(~	

Figure 5. 31 Your apps section in General Tab

Enter the Android package name from the Android Studio project, which contains the Diet Manager app; in this case should be: com.example.crudfirestore, then click on "Register app" to continue.



i igure 5. 52 i ida i indioia app page ste

Download the google-services.json file,



Figure 5. 33 Add Android app page Step 2

And move google-services.json file into the Android Studio module (app-level) root directory. Which will be:

 $... \verb| crudfirestore \verb| android \verb| app \verb| google-services.json|| \\$



Figure 5. 34 google-services.json location in the Android Studio project file directory

Then ensure to have the Firebase SDK is set-up according to the provided steps: (The provided Android Studio project files containing the Diet Manager have implemented the following steps)

```
Add Firebase SDK
Instructions for Gradle | Unity 🛛 C++ 🛛
1. To make the google-services.json config values accessible to Firebase SDKs, you need
 the Google services Gradle plugin.
 Add the plugin as a buildscript dependency to your project-level build.gradle file:
 Root-level (project-level) Gradle file (<project>/build.gradle):
    buildscript {
       repositories {
         // Make sure that you have the following two repositories
         google() // Google's Maven repository
                                                                             6
         mavenCentral() // Maven Central repository
       dependencies {
         // Add the dependency for the Google services Gradle plugin
                                                                             Ū
         classpath 'com.google.gms:google-services:4.3.15'
    allprojects {
       repositories {
        // Make sure that you have the following two repositories
google() // Google's Maven repository
                                                                             mavenCentral() // Maven Central repository
    }
```



2. Then, in your **module (app-level)** build.gradle file, add both the google-services plugin and any Firebase SDKs that you want to use in your app:

Module (app-level) Gradle file (<project>/<app-module>/build.gradle):



Figure 5. 36 Add Android app page Step 3 (part 2)

5.3.5 Mobile Device Set-up

After the mobile device is connected and detected by the computer successfully, enable Developer Options on the mobile phone, and check the mobile device manufacturer for the steps.

When Developer Options are enabled, navigate to Settings -> Developer Options -> Enable USB Debugging.



Figure 5. 37 Developer options screen in mobile device's Settings

Then on the USB Settings on the mobile phones, ensure that the USB is connected using the "USB tethering" mode.



Figure 5. 38 USB Settings screen

The device will be connected successfully for the Android Studio to perform running and testing the Diet Manager app.

The connected device will be detected by the Android Studio and will be shown on the top of the screen.

-	<u> E</u> ile <u>E</u> dit <u>V</u> iew <u>N</u> avigate <u>C</u> od		<u>R</u> efactor <u>B</u> uild R <u>u</u> n <u>T</u> ools	VC <u>S</u> <u>W</u> indow	Help crudfirestore [\crudfirestore] - main							
сп	udfirestore 👌 lib 👌 🛃 main.dart		🚦 SM N770F (mobile) 🔻 🗌	🎸 main.dart 🤊		R	š, 🔳	Q.	₩ 1	Q	¢ I	
E.	⊕ Ξ ÷ 🗢 –	1	Chrome (web)									4
Iag	er settings.gradle		📘 Edge (web)									
Mar					if i a course a contradice i							
8	assets		Windows (desktop)		abase/firebase_database.dart';							r In
10	> 🛄 ios				rial.dart';							sbe
Ses	Y 🖿 lib		Open Android Emulator: S	Samung Clone								
	> 🖿 auth page		Pefrerh									
	· · · · · · · · · · · · · · · · · · ·											
+	main_page		import 'auth pag	e/authpage.	dart':							G
<u>e</u>	📸 chartdata.dart											
- O'	A second seco		innoort 'auth nac	<u>le/utils.dar</u>								

Figure 5. 39 Connected device at the top of the screen of Android Studio

5.4 System Operations

In this subsection, all Diet Manager's operations and user interactions with the app were discussed. The operations will include User Authentication, such as Sign-Up user for a new account, Sign-In user account, Reset the User Password, and Page Navigations operations that navigate the user to all functions of the app.

As well as the major functions of the app were discussed, consisting of all the six subobjectives accordingly of the project, which are the Record Page functions, Progress Page functions, Reminder Page function, Timelapse Page function and Streak Page function. The other functions, such as the User Profile Page function and Welcome Page function, will be discussed as well.

5.4.1 User Authentication

i. Sign-In

The Sign In screen allows user to enter their Email and Password and click "Sign in" to sign-in to their Diet Manager account, and then it will navigate to the Diet Manager app. The password field has a hide and unhide password to view or hide the entered password. The screen has Forgot Password and Sign Up text link that will navigate to the Reset Password Page and Sign Up page.

	diallo
Diet Manager	
Diet Manager	
🖂 Email	
A Password	0
Sign In	
Forgot Password?	
Don't have an account? Sign Up	

Figure 5. 40 Diet Manager's Sign In screen

ii. Sign-Up

The Sign-up screen allows user to enter their Email and Password and click "Sign Up" to create a Diet Manager account, and then it will navigate to the Diet Manager app. The password field has a hide and unhide password to view or hide the entered password. The screen has Log In text link that will navigate to the Sign In page.

	Disilio
Create an account	
Enter your email and password to sign up.	
🖂 Email	
Password	Ο
Sign Up	
Already have an account? Log In	

Figure 5. 41 Diet Manager's Sign Up screen

After the user has successfully Sign-Up for an account, the user will be navigated to User Profile Page to set up their user profile by collecting the user's personal information to create their diet plan. (Refer to Chapter 5.4.9 for details on User Profile Page)

iii. Reset User Password

The Reset Password screen enables users to enter their account email and click "Reset Password" to send an email to reset their account password.



Figure 5. 42 Diet Manager's Reset Password screen

The Reset Password will be emailed to the user.



Figure 5. 43 Reset Password email in Inbox

Clicking on the link will navigate the user to the webpage to enter their new password.

Users will need to enter their new password and click on the "SAVE" button to change to a new password.



Figure 5. 44 Reset Password webpage

5.4.2 Page Navigation

The Diet Manager app has a Bottom Navigation Bar for navigating between the main functionality page of the app and a Navigation Drawer on the right side of the screen for navigating between other functionalities pages.

i. Bottom Navigation Bar

The Bottom Navigation Bar is displayed automatically at the bottom screen of the app. It contains four tabs in the bottom navigation bar, which are Record, Progress, Timelapse, and Streak, with each tab having its own distinctive icon to represent each tab.

Tapping on any of the tabs will navigate the user to the page, and the selected tab will display the name of the tab next to the icon when the user is viewing the current page.



Figure 5. 45 Diet Manager's Bottom Navigation Bar

ii. Navigation Drawer

The Navigation Drawer will be displayed when the user taps on the blue Profile Icon located at the top right corner of the screen in the app.



Figure 5. 46 Diet Manager's Profile Icon

The Navigation Drawer will be displayed on the right side of the screen. The navigation drawer contains a profile icon with the username, and four navigation tabs, which are Profile, Reminder, How It Works, and Sign Out.



Figure 5. 47 Diet Manager's Navigation Drawer

5.4.3 Record Page

The Record Page is the first screen displayed to the user when the user launches the app. The Record Page is the first sub-objectives of the project, which is to calculate the user's food calorie intake in every meal. (Refer to Chapter 1.2, Sub-objectives 1)

Before the user is able to begin using the Diet Manager's app functionalities, the user's personal information, such as age, gender, current weight, target weight goal, physical activity level and starting diet date, will be collected after signing up for a new user account to be used as input to produce Remaining Calories, and Daily Calories Limit for the Record Page. (Refer to Chapter 5.4.9 for further details on collected user personal information)

Record Page Details	Record Page Screen
The Record Page consists of a few crucial	Record Your Weight
key elements:	Today's Romaining Calorias
	1
(refer to the numbering on the screen)	689 kcal 📃
	2
1. Remaining calories intake for the day,	Total Calories Consumed: 603 KCCI
which displays the limited amount of	603 / 1292 kcal
calories that the user can consume for	View Calories History
the day to diet.	5
2. Total calories consumed for the day,	3
which displays the sum of all the	Apples 52 kcal
calories of the added food items in the	SZ RCUI
list below.	
3. List of consumed food and its calories,	Bread White 119 kcal
which displays all the user-added food	
items for the day.	Lowfat Chocolate
4. Add a food button for the user to add	Milk I52 KCCI
food items to list	
5. View Calories History to view all past	sausage Pizza 280 kcal
records of consumed food.	
6. Progress Bar to track the amount of	4
calories consumed against the total	↑ Record ** IIE 6
calorie limit for the day.	
	Figure 5. 48 Diet Manager's Record Page


The selected food calories will appear automatically in the Calories text fields. By default, the number of Calories is set by default with a weight of 100g.	Add food item Food Apples Calories 52 Quantity (g) Save Cancel Figure 5. 50 Calories values in Add food item dialogue box
Users can specify the quantity of the food in the units of grams, and the value in the Calories fields will be adjusted automatically to the specified amount of calories entered. Users can click on the "Save" button to add the food item.	Add food item Food Apples Calories 26.0 Quantity (g) 50 Save Cancel Figure 5. 51 Quantity values in Add food item dialogue box





	Food History List
The View Calories History text, when tapped, will navigate the user to the Food History List page, which displays all past	2023-4-18 603 kcal
recorded consumed food and calories by date.	2023-4-17 990 kcal
	2023-4-16 1028 kcal
	2023-4-15 1125 kcal
	2023-4-14 1054 kcal
	2023-4-13 1052 kcal
	2023-4-12 732 kcal
	2023-4-11 859 kcal
	Figure 5. 56 Food History List screen

	2023-4-18
Osers can tap on any of the dates, and it will navigate the user to a Food History Details page that displays all the consumed foods	Total
for the selected day.	Calories:
	603 kcal
	Foods:
	Apples 52 kcal
	Bread White Wheat 119 kcal
	Lowfat Chocolate Milk 152 kcal
	Sausage Pizza 280 kcal
	Figure 5. 57 Food History Details screen

Table 5. 1 Record Page Screen and Details

5.4.4 Progress Page

The Progress Page consisted of the second and third sub-objectives of the project, which is to monitor user dieting progress to keep track of user diet goals. (Refer to Chapter 1.2, Sub-objectives 2 & 3)

Before the user is able to begin using the Diet Manager's app functionalities, the user's personal information, such as age, gender, current weight, target weight goal, physical activity level and starting diet date, will be collected after signing up for a new user account to be used as input to produce Target Weight, Weight Goal and Weight Goal

CHAPTER 5

Date for the Progress Page. (Refer to Chapter 5.4.9 for further details on collected user personal information)









 Table 5. 2 Progress Page Screen and Details

5.4.5 Welcome Page

The Welcome Page is crucial for the Diet Manager app. The Welcome Page will be displayed to the user right after the user has signed up for a new account. And the Welcome Page can be accessed again in the Navigation Drawer for the user to view and update the user details. (Refer to Chapter 5.4.2 for details on accessing the Welcome Page)

Welcome Page Details	Welcome Page Screen
----------------------	---------------------









Table 5. 3 Welcome Page Screen and Details

5.4.6 Reminder Page

The Reminder Page is the fourth sub-objectives of the project, which is to remind the user to input food calorie intake when the user forgets to. (Refer to Chapter 1.2, Sub-objectives 4)

Reminder Page Details	Reminder Page Screen

The Reminder Page consists of a few crucial	Reminder	
 The Reminder Page consists of a few crucial key elements: (refer to the numbering on the screen) 1. Edit Time button, for the user set a time for daily reminders. 2. Notification On toggle, to enable or disable 	Reminder Steate a reminder to remind yourself daily to ensure you always stay on track on your diet goals! 8:77 PM Edit Time Notification On 2	
	Figure 5. 67 Diet Manager's Reminder screen	



 Table 5. 4 Reminder Page Screen and Details

5.4.7 Timelapse Page

The Timelapse Page is the fifth sub-objectives of the project, which is to develop a timelapse of the user's diet progress using pictures that the user saves daily for the user to have an overview of their diet progress. (Refer to Chapter 1.2, Sub-objectives 5)











Table 5. 5 Timelaspe Page Screen and Details

5.4.8 Streak Page

The Streak Page is the sixth sub-objectives of the project which is to collect streak points when user did not exceed their daily calories intake limit. (Refer Chapter 1.2, Sub-objectives 2)



CHAPTER 5











Table 5. 6 Streak Page Screen and Details

5.4.9 User Profile Page

The User Profile Page is the most crucial page in the Diet Manager app. The User Profile Page will be displayed to user right after used had signed up for a new account. And the User Profile Page can be accessed again in the Navigation Drawer for user to view and update the user details. (Refer to Chapter 5.4.2, for details on accessing the User Profile Page)

The User Profile Page has the following main functions:

- 1. Collect user's personal information such as age, gender, weight, target weight etc.
- 2. Create personalised diet plan based on collect user information,
- 3. Calculate and produce data for other functions in the app such as Record Page, and Progress Page. (Refer to Chapter 5.4.3 & 5.4.3, for details on user data used)

User Profile Page Details User Profile Page Screen		
The User Profile Page consists of few	User Details	
 crucial key elements: (refer numbering on the screen) 1. User Diet Goal dialog, a simple SMART diet goal created for user to get an overview of their dieting journey. 	Hi, I'm Chee Wah, 22 years old, male, and my current weight is 55.0 kg. I aim to achieve 50.0 kg within 70 days by limiting my calories intake to 1518 kcal daily. My diet journey begins on 19/4/2023, and this is my story!	
2. Name, for user to enter their name	Name	
3. Gender, for user to select gender	Chee Wah	
4. Age, for user to enter their age		
5. Height, for user to enter their height		
 Current Weight, for user enter their current weight Target Weight Goal, for user to enter 	Age 4 22 4 Height 5	
their targeted weight goal	Surrent Weight	
8. Physical Activity Level, for user to select their Physical Activity Level	55.0 6	
 Starting Diet Date, for user to select the starting diet day 	50.0 7	
10. Save Button, to save all user details.	Figure 5. 84 Diet Manager's User Details screen	

User Diet Goal dialog, a simple SMART diet goal created for user to get an overview of their dieting journey.

The details which in bold, such as Name, Gender, Age, Current Weight, Target Weight Goal and Physical Activity Level (PAL) in the dialog summary are obtained from the calculation of all the user entered details in the text boxes at the second half of the screen. (Refer to the last part of the subsection to for details of the Harris–Benedict calculation)

Then the dialog summary will produce 2 important details which is the daily calories intake limit and the number of days to achieve the targeted weight goals.

The details in the dialog are constantly being updated as user entering their details in the text boxes.

User Details

Hi, I'm Chee Wah, 25 years old, male, and my current weight is 55.0 kg. I aim to achieve 50.0 kg within 70 days by limiting my calories intake to 1268 kcal daily. My diet journey begins on 19/4/2023, and this is my story!

Figure 5. 85 SMART Diet Goal dialog

The number of days to achieve diet weight goal and daily calories intake limit are produces by considers of the user's age, gender weight, target weight goal, and Physical Activity Level (PAL) from the user input text boxes.

It then calculates the user's Basal Metabolic Rate (BMR) using the Harris-Benedict equation. Then, using the user's PAL and BMR to determine the user's daily calorie intake limit.

The number of days needed for the user to achieve their target weight goal were calculated using the differences between the weight and target weight goal and assuming a healthy weekly weight loss of 500g.

The user can click on the blue disk card icon to Save the changes and details.

The updated details will be used by other functions by the app to perform diet functionalities such as Record Page, and Progress Page. (Refer to Chapter 5.4.3 and 5.4.4)



Figure 5. 86 User Profile's user details used for calculation



Figure 5. 87 User Profile's Save Icon button

The Harris–Benedict equations revised by Roza and Shizgal in 1984 [5], the equations are as follows:

To determine the daily caloric intake limit for weight loss, first calculate the Total Daily Energy Expenditure (TDEE) using the Basal Metabolic Rate (BMR) and Physical Activity Level (PAL). After that, calculate the daily caloric intake needed to lose weight by creating a calorie deficit.

The procedures were outlined below in detail:

1. Calculate BMR:

For men:

 $BMR = 88.362 + (13.397 \times weight in kg) + (4.799 \times height in cm) - (5.677 \times age in years)$

For women:

 $BMR = 447.593 + (9.247 \times weight in kg) + (3.098 \times height in cm) - (4.330 \times age in years)$

Calculate TDEE by multiplying BMR by PAL:
 TDEE = BMR x PAL

Below displayed the value for PAL according to the user's exercise frequency:

PAL = 1.2

Sedentary (little or no exercise): This level represents people with a desk job or those who don't engage in any physical activity during the day.

PAL = 1.375

Lightly active (light exercise or sports 1-3 days/week): This level is for individuals who perform light exercise or engage in sports a few times a week.

PAL = 1.55

Moderately active (moderate exercise or sports 3-5 days/week): This level is for those who engage in moderate exercise or sports for the majority of the week.

PAL = 1.725

Very active (hard exercise or sports 6-7 days/week): This level is for people who engage in intense exercise or sports almost every day of the week.

PAL = 1.9

Extremely active (very hard exercise or sports, physical job, training twice a day): This level is for individuals who perform very intense exercise or sports, have a physically demanding job, or train multiple times a day.

3. Make a calorie deficit to lose weight. A typical strategy is to reduce calories by 500 per day, which could lead to weekly weight loss of about 500g:

5.5 Implementation Issues and Challenges

This section discussed the issues and challenges faced during the implementation of all the functions in the Diet Manager project and the solutions on how each issue was weight and taken into consideration to overcome each of the challenges.

1. User-friendly and intuitive UI/UX

The app must have a user-friendly user interface that is simple to use and navigate. It could be difficult to create an intuitive interface that satisfies all the different demands and preferences of people. Layout, typography, colour schemes, and user interactions all need to be carefully considered when designing an interface that is both visually appealing and easy to use.

Another crucial aspect is responsiveness, which makes sure that the user interface of the app adjusts to various screens and devices without any difficulty. The use of a library like Syncfusion, which offers a scalable and adaptable charting solution, should be taken into account while building the weight chart, for instance. To ensure that users understand the data in the chart, it is also crucial to provide visual cues like tooltips or labels.

2. Integration with Firebase:

Integration with Firebase: Using the appropriate SDKs and having a thorough grasp of the documentation is essential for integrating Firebase services. For instance, managing different authentication states like sign in, sign up, and sign out and making sure the UI updates appropriately are required when integrating Firebase Authentication. Firestore may have difficulties when it comes to effectively organising the data, creating the database schema, and utilising the right queries to access and update the data. Understanding the data structure as a JSON tree and managing real-time updates might be difficult for Realtime Databases. For example, it is critical to take this into account while handling various circumstances, such as account creation, password recovery, and email verification, when implementing user authentication.

3. Calculation accuracy

The numerous factors that must be considered when determining the maximum daily caloric intake and the number of days needed to attain the desired weight. For example, the basal metabolic rate (BMR), a key component in determining the daily calorie intake limit, must be calculated using complicated calculations like the Harris-Benedict equation. Calculations were further complicated by the necessity to account for the user's age, weight, gender, and degree of exercise. Additionally, it must be sure that the calculations are correct and dependable regardless of the measuring unit used by carefully considering the measurement units used in the app, such as metric and imperial.

The success of a health and fitness app depends on the implementation of precise and reliable complex calculations; thus, attention must be taken to guarantee that these calculations are done correctly.

5.6 Concluding Remarks

To summarize this chapter, the software setup for the Diet Manager app was discussed. The software setup includes installing the Flutter SDK, Java JDK 11, Android Studio, and Firebase. Additionally, connecting the mobile device to the computer using a USB cable is also required, and in some cases, installing the appropriate USB driver for the mobile device. Then, the details on the settings and configurations required for each component of the software setup are discussed as well.

Then the next section describes the various operations and user interactions in the Diet Manager app. The User Authentication process includes Sign-In, Sign-Up, and Resetting User Password. Page Navigation is done through the Bottom Navigation Bar and the Navigation Drawer. The Record Page calculates daily food calorie intake, while the Progress Page monitors the user's dieting progress. The Welcome Page provides an explanation of the Diet Manager app, including the Harris-Benedict equation used for weight loss. The Reminder Page reminds users to input food calorie intake, and the Timelapse Page lets users create a timelapse of their diet progress. The Streak Page in the Diet Manager app allows users to collect streak points for not exceeding their daily calorie limit. The User Profile Page is the most important page and allows users to enter their personal information and create a personalized diet plan. The page calculates the

daily calorie intake limit based on user data using the Harris-Benedict equation and creates a daily calorie deficit to lose weight. The TDEE is calculated by multiplying the BMR with PAL, and a typical strategy is to reduce 500 calories per day for a weekly weight loss of 500g.

Then the implementation and challenges faced are the implementation of a user-friendly UI/UX, integration with Firebase, and accurate calculations are important considerations for building a successful health and fitness app. The UI/UX must be intuitive and responsive, and Firebase integration requires careful attention to authentication and database management. Calculating daily caloric intake and weight loss targets accurately involves complicated equations, measurement unit conversions, and careful consideration of various user factors. Precise and reliable calculations are crucial for the success of a health and fitness app.

In conclusion, the Diet Manager app provides a comprehensive solution for managing daily food intake and weight loss goals. From the software setup to the various app functionalities, every aspect is designed to help users achieve their dieting objectives. However, building a successful health and fitness app requires addressing challenges such as creating an intuitive UI/UX, integrating with Firebase, and ensuring accurate calculations. By considering these factors, we can build a reliable and effective app that helps users maintain a diet management app.

CHAPTER 6 SYSTEM EVALUATION AND DISCUSSION

6.1 System Testing and Performance Metrics

This section provides an overview of the system testing performed on the application, focusing on integration testing techniques and their role in ensuring seamless functionality between different components. The section also covers the performance metrics used to evaluate the effectiveness of the testing process.

The primary testing technique used in the application is integration testing, which focuses on verifying the functionality of different components when they work together. It helps identify any issues or inconsistencies that may arise when individual modules or components interact with each other. Integration testing is essential in this context because it ensures that the various functions of the Diet Manager application, such as the Record Page, Progress Page, Reminder Page, Timelapse Page, Streak Page, and User Profile Page, work together as intended, providing a smooth and consistent user experience. (Refer to Chapter 6.2 for details of the testing setup and results for each of the functions mentioned earlier)

To evaluate the effectiveness of the integration testing process, various performance metrics are used, including:

- 1. Functionality: Ensuring that each component of the application performs its intended function and that data flows correctly between components.
- 2. User experience: Verifying that user actions are correctly processed and lead to the expected outcomes within the system, such as inputting data, navigating through the app, and manipulating stored data.
- 3. Data accuracy: Checking that calculations and data processing within the application are accurate and consistent, such as in the case of the Record Page, where adding a food item updates the Remaining Calories Intake, Total Calories Consumed, and Progress Bar.

- 4. Reliability: Assessing the application's ability to handle different user inputs and scenarios without errors or unexpected behaviour.
- 5. Efficiency: Measuring the application's performance in terms of response times, resource usage, and overall system load.

6.2 Testing Setup and Result

The Integration testing will be performed on only the major functions of the Diet Managers. The major functions, which are the Record Page, Progress Page, Reminder Page, Timelapse Page and Streak Page, consist of all the six sub-objectives accordingly of the project. And lastly, the other tested function included was the User Profile Page. The result of each test performed on each of the functions will be summarized.

The Test Objectives for each of the functions were summarized as follows:

- 1. Record Page: Verify the accuracy and functionality of daily calorie intake calculations, food item addition and removal, and navigation to the Calories History page.
- 2. Progress Page: Validate the display and update of target weight, weight goal date, progress bar calculations, and weight record management.
- 3. Reminder Page: Verify the functionality of setting, activating, and deactivating reminder notifications.
- 4. Timelapse Page: Test photo saving and display, photo selection, and timelapse video creation.
- 5. Streak Page: Validate the display and management of streak scores, including streak redemption logic.
- 6. User Profile Page: Verify accurate diet calculations based on user inputs, information display, and data storage.

6.2.1 Record Page

Integrated Testing 1: Record Page

Test Objective: To test the Daily Calories Intake limit, Remaining Calories Intake, Total Calories Consumed and added food items were displayed correctly, as well as the Add food button and the View Calories History button able to function accordingly.

Input	Expected Output	Results
Check the value of the Daily Calories Intake Limit in the Progress Bar.	The value of the Daily Calories Intake Limit in the Progress Bar is similar and accurate to the value of the Daily Calories Intake Limit in the User Profile Page. (Refer to Chapter 5.4.9 for details of the Daily Calories Limit of the User Profile Page)	PASS
Click the " + " blue icon to add food items.	The "Add food item" dialogue is displayed with all the Food, Calories and Quantity textboxes empty, as well as the "Save" and "Cancel" button.	PASS
Click on the "Sour"	The inputted value in the Food, Calories and Quantity textboxes in the dialogue box closed after the "Save" button is clicked. And the Food item will be added and displayed along with the food name and its calorie values in the food list in the Record Page.	PASS
button after inputting into the Food, Calories and Quantity textboxes in the Add Food item dialogue box.	The Remaining Calories Intake will be updated to deduct the added food item calories in the food list when a new food item is added.	PASS
	The Total Calories Consumed will be updated, to sum up the added food item calories in the food list when a new food item is added.	PASS
	The Progress Bar will be filled up corresponds to the sum of the amount of added food item calories in the food list and displays the value of Total Calories Consumed over the value of Daily Calories Intake Limit.	PASS
Swipe the Food Item either to the left or right in the food list.	The Food Item will be deleted and removed from the food list on the Record Page.	PASS
--	--	------
	The Remaining Calories Intake will be updated to add the removed food item calories to the food list when a food item is removed.	PASS
	The Total Calories Consumed will be updated to recalculate and sum all food item calories that exist in the food list when the food item is removed.	PASS
	The Progress Bar will be updated and filled up corresponds to the sum of the amount of food item calories that exist in the food list and displays the value of Total Calories Consumed over the value of Daily Calories Intake Limit.	PASS
Click on the "View Calories History"	Navigates user to a Calories History page displaying all the past calories consumed arranged in a list by dates.	PASS
Click on any of the dates items on the Calories History page	Navigates user to a Food Details page displaying all the foods and their calories consumed for the selected day.	PASS

Table 6. 1 Integration Testing for Record Page Functions

Results: All tests passed, indicating that the system is displaying the correct values and providing accurate functionality for adding and removing food items, updating calorie counts, and navigating to the Calories History page.

6.2.2 Progress Page

Integrated Testing 1: Progress Page

Test Objective: To test the value of the Target Weight, Weight Goal Date, Progress Bar weight to lose and added Weight items are displayed correctly on the Progress Page. And to check if the weight record value plotted on the graph is respective to the weight record item in the weight list.

Input	Expected Output	Results
Check the value of the Target Weight.	The value of the Target Weight is similar and accurate to the value of the Target Weight set on the User Profile Page. (Refer to Chapter 5.4.9 for details of the Target Weight on the User Profile Page)	PASS
Check the date of the Weight Goal Date.	The date of the Weight Goal Date is similar and accurate to the sum of days to reach the target weight goal set in the User Profile Page and the current date. (Refer to Chapter 5.4.9 for details of the days to reach the target weight goal on the User Profile Page)	PASS
Check the value of the weights to lose in the Progress Bar.	The value of weight to lose in the Progress Bar is similar and accurate to the difference between the target weight goal and the current weight set in the User Profile Page. (Refer to Chapter 5.4.9 for details of the days to reach the target weight goal on the User Profile Page)	PASS
Click the "+" blue icon to add a new weight record.	The "Add a new weight" dialogue is displayed with all the Weight textboxes empty, as well as the "Add Weight" button.	PASS
Click on the "Add weight" button after inputting into the Weight textbox in	The weight record will be added and displayed along with the weights and dates in the weight list on the Progress Page.	PASS
the "Add a new weight" dialogue box.	The Weight Goal Date will be updated to recalculate the number of days to reach the diet weight goal based on the latest weight recorded added.	PASS

	The Progress Bar will be updated with the amount of weight loss based on the added latest weight record minus the target weight.	PASS
	The Progress Bar will be filled up corresponding to the amount of weight loss and displays the value of weight lost over the number of weights required to lose.	PASS
	The Chart graph will be updated to add a new point of the latest added weight record and its date.	PASS
	The Chart graph will be updated to add the target weight goal point weight record and its date based on the latest weight record.	PASS
Swipe the Weight Record item either to the left or right in the food list.	The Weight record item will be deleted and removed from the weight list on the Progress Page.	PASS
	The Weight Goal Date will be updated to recalculate the number of days to reach the diet weight goal based on the latest weight recorded available.	PASS
	The Progress Bar will be updated with the amount of weight lost based on the available latest weight record minus the target weight.	PASS
	The Progress Bar will be updated to correspond to the amount of weight lost and displays the value of weight lost over the number of weights required to lose.	PASS
	The Chart graph will be updated to the point of the available existing weight record and its date.	PASS
	The Chart graph will be updated to add the target weight goal point weight record and its date based on the latest available weight record.	PASS

Click on any of the eight points in the Graph.	A small box will be displayed on top if the clicked point to display the weight and its date plotted.	PASS
--	--	------

Table 6. 2 Integration Testing for Progress Page Functions

Results: All tests passed, showing that the system is accurately displaying target weight, weight goal date, and progress metrics, as well as correctly updating these values based on added or removed weight records.

CHAPTER 6

6.2.3 Reminder Page

Integrated Testing 1: Reminder Page

Test Objective: To test the reminder time were able to set time and turn on or off the notification.

Input	Expected Output	Results
Click the "Pick	Display a Time Picker for the user to select a reminder time	PASS
Time" button	Display the selected reminder time on the Reminder page	PASS
Toggle the "OFF" button of the "Notification On."	Deactivate the reminder notification set on the selected reminder time; the notification will not pop up even if the current time has reached the selected reminder time set by the user.	PASS
Toggle the "ON" button of the "Notification On."	Activates the notification settings on the selected reminder time, a reminder notification will pop up when the current time has reached the selected reminder time set by the user	PASS

Table 6. 3 Integration Testing for Reminder Page Functions

Results: All tests passed, demonstrating that the system effectively sets and activates/deactivates reminder notifications as desired.

6.2.4 Timelapse Page

Integrated Testing 1: Timelapse Page

Test Objective: To test whether the photos taken by the user were able to be saved and displayed correctly on the screen, select all photos function able, selected photos are able to be downloaded and create a timelapse video.

Input	Expected Output	Results
Click on the "Camera" icon button.	Navigates user to built-in camera function to take photos	PASS
Take a picture with the built-in camera.	The photo taken is saved and displayed in a grid view in the gallery of the Timelapse page.	PASS
Click on any of the pictures in the gallery.	Navigate the user to a new page with the clicked pictures enlarged to fullscreen the pictures.	PASS
Long-click on any of the pictures in the gallery.	The selection mode will be activated. Selected photos will be shown a ticked icon to indicate picture has been selected.	PASS
Clicking on the "Select All" square icon button	The selection mode will be activated, and all pictures in the gallery will be selected. The selected picture can be selected or unselected by clicking once for either option.	PASS
Clicking on the "X " red icon button while selection mode active	on the "X on button Deactivate the selection mode, and unselect all pictures selection in the gallery. ve	
Clicking on the "Download" icon button while selection mode active	Brings users to browse on mobile storage locations to save the selected photos. Then ask for user phone permission for file storage before the selected pictures are downloaded and stored in the mobile storage.	PASS

Bachelor of Information Systems (Honours) Information Systems Engineering Faculty of Information and Communication Technology (Kampar Campus), UTAR

Clicking on the	Brings users to browse mobile storage locations to save	
	the timelapse video based on the selected pictures. Then	
"Video" icon button	ask for user phone permission for file storage before the	PASS
while selection	salaatad nicturas wara usad to graata timalansa video and	
mode active	selected pictures were used to create timerapse video and	
mode active	were stored in the mobile storage.	

Table 6. 4 Integration Testing for Timelapse Page Functions

Results: All tests passed, indicating that the Timelapse Page properly saves and displays photos and allows users to create timelapse videos from selected images.

6.2.5 Streak Page

Integrated Testing 1: Streak Page

Test Objective: To test the value of the Current Streak, the Highest Streak Scores were saved and displayed correctly. And to test the streak redemption logic.

Input	Expected Output	Results
Click on the "Fire" icon button when the user's daily calories are within the limit.	The "Redeem Streak" will appear, and upon clicking the "Redeem" button, the Streak has been redeemed, and the Streak count will increment by 1.	PASS
Click on the "Fire" icon button when the user has already redeemed the streak for the day.	The "Redeem Streak" will appear, and upon clicking the "Redeem" button, the Streak will fail to be redeemed, and the "Failed" dialogue will appear informing the user have redeemed the streak for the day, the existing Streak count will remain unchanged.	PASS
The user's daily calories limit was exceeded for the day	The "Failed" dialogue will appear automatically to inform that the user's daily calorie limit has been exceeded and 5 Streak points have been deducted automatically as penalties.	PASS
The user missed a day redeeming their Streak.	The "Failed" dialogue will appear automatically to inform that the user missed redeeming their streak, and 5 Streak points have been deducted automatically as penalties.	PASS
The Current Streak count is higher than the Highest Streak Score	The Highest Streak Score will be automatically be updated to the current streak value.	PASS

Table 6. 5 Integration Testing for Streak Page Functions

Results: All tests passed, showing that the Streak Page accurately displays streak values and applies the appropriate streak redemption logic.

6.2.6 User Profile Page

Integrated Testing 1: User Profile Page

Test Objective: To test the user details that were inputted by the user in the textbox of the User Profile Page were performing the diet calculation accurately, displaying the calculated information correctly, as well as storing and loading the saved user details correctly in the User Profile Page.

Input	Expected Output	Results
Input all user details in text boxes. The text boxes are Name, Gender, Age, Current Weight, Target	The textbox displayed all users' inputs and the user details textboxes correctly.	PASS
Weight Goal, Physical Activity Level, and Starting Diet Date.	The summary dialogue box displayed all users' inputs and the user details textboxes correctly.	PASS
The input value for diet calculation in user details textboxes. Textbox value: Gender = Male Age = 22	Calculate and display the output for the daily calorie intake limit and the number of days to reach the targeted weight goal correctly using the correct calculation in the summary dialogue box. Expected Output Calculation: (Refer to Chapter 5.4.9 for Diet Calculation) $BMR = 88.36 + (13.375 \times 55) + (4.8 \times 160) - (5.7 \times 22) = 1467.96$	PASS
 Height = 160 Current Weight = 55 Weight Goal = 50 PAL = 1.375 	TDEE = 1467.96 x 1.375 = 2018.45 Daily Calories Intake Limit = 2018.445 - 500 = 1518.45 kcal No. Days to Reach Goal = (55-50)/0.5 * 7 = 70 days	
Select the "Save" blue icon button to save all the inputted details.	The saved user details load and displays into the textboxes correctly after closing and reopening the page.	PASS

Table 6. 6 Integration Testing for User Profile Page Functions Bachelor of Information Systems (Honours) Information Systems Engineering Faculty of Information and Communication Technology (Kampar Campus), UTAR Results: All tests passed, demonstrating that the User Profile Page correctly calculates and displays diet information and properly stores and loads user details.

6.3 Project Challenges

This section depicts some of the major challenges faced during the development of the Diet Manager project and how each challenge was weight and taken into consideration to overcome each of the challenge.

1. UI Design challenges:

The most important aspect of any app's success is designing an interface that is both aesthetically pleasing and easy to use. The UI for the Diet Manager app must present the relevant details and data in an orderly and simple way.

Here are some of the difficulties with the Diet Manager app's user interface design faced:

- **Displaying a lot of data:** The app must show users' diet progress, as well as other details such as a lot of users, added weight record lists, graph charts with weight points plotted, consumed calorie lists, and other elements. It is important to have users be able to discover and get the information they require with ease, thanks to the UI's design.
- Consistency in design and layout: The usability of an app depends on its consistency in each of these areas. The app's UI components, including the colours, fonts, and layout, were set to be uniform throughout. It makes it easy for the user's eye to browse and navigate around the app.
- Developing an intuitive navigation system: The navigation system, such as the bottom navigation bar and navigation drawer implemented, must be user-friendly and enables users to simply switch between its many pages and functions.

2. Optimizing Performance Challenges:

To guarantee that the Diet Manager app operates quickly and effectively on all devices, performance optimization is essential. For the Diet Manager app, these issues were presented when optimizing performance:

- Storing and managing a lot of data: Including user information and food databases will be necessary for the app. The data storage and retrieval system for the app, such as the Firebase Firestore database, must be designed for optimization for it to function swiftly and effectively.
- Minimizing load times: To give users a smooth experience, the Diet Manager's pages must load quickly. By resizing and compressing pictures taken by the user before uploading and storing them to Firebase Storage in the Timelapse pages, load times can be reduced significantly.
- Animation optimization: Animations can enhance the Diet Manager's aesthetic appeal but, if not done correctly, can also make it less responsive. Animations were used with caution and minimized so that they perform main functions and load multiple element's animations without lengthening the app's screen loading time.

6.4 Objectives Evaluation

This section discussed the evaluation of each objective of the project and how it has been achieved.

6.4.1 Evaluation of Sub-objective 1

The sub-objective is to develop a diet management module to calculate the user's food calorie intake in every meal. (Refer to Chapter 1.2, Sub-objectives 1)

Evaluation: The diet management module has been successfully developed and tested. The system accurately calculates calorie intake for every meal and updates the user's total daily calorie consumption. This objective has been fulfilled.

6.4.2 Evaluation of Sub-objective 2

The sub-objective two is to develop a weight tracking tool to record users' weight and for users to keep track of their weight. (Refer to Chapter 1.2, Sub-objectives 2)

Evaluation: The weight-tracking tool has been successfully implemented and tested. The Progress Page provides users to record their weights and display them in a graph for users to visualize and keep track of their weight progress.

6.4.3 Evaluation of Sub-objective 3

Sub-objective three is to develop a summary report system to monitor user dieting progress to keep track of user diet goals. (Refer to Chapter 1.2, Sub-objectives 3)

Evaluation: The summary report system has been implemented and tested. The Progress Page provides a clear visualization of the user's diet goals, dates, and weight loss progress and visualizes their diet goal with a line graph. This objective has been fulfilled.

6.4.4 Evaluation of Sub-objective 4

Sub-objective four is to develop a reminder tool for users to input user's food calorie intake when the users forget to. (Refer to Chapter 1.2, Sub-objectives 4)

Evaluation: The Reminder Page has been developed and tested, successfully allowing users to set and manage reminder notifications for inputting food calorie intake. This objective has been fulfilled.

6.4.5 Evaluation of Sub-objective 5

Sub-objective five is to develop a timelapse of the user's diet progress using pictures that the user saves daily for the user to have an overview of their diet progress. (Refer to Chapter 1.2, Sub-objectives 5)

Evaluation: The Timelapse Page has been implemented and tested, enabling users to save daily pictures and create a timelapse video showcasing their diet progress. This objective has been fulfilled.

6.4.6 Evaluation of Sub-objective 6

Sub-objective six is to develop a streak system for users to collect streak points when the user does not exceed their daily calorie intake limit. (Refer to Chapter 1.2, Subobjectives 6)

Evaluation: The Streak Page has been successfully developed and tested. Users can now collect streak points for staying within their daily calorie limits, thus encouraging adherence to their diet goals. This objective has been fulfilled.

6.5 Concluding Remark

To summarize, this Chapter has discussed system testing and performance metrics used in an application, with a focus on integration testing. Integration testing is essential for ensuring that various components of the application work together seamlessly, providing a consistent user experience. Performance metrics used to evaluate the testing process include functionality, user experience, data accuracy, reliability, and efficiency. Integration testing focuses on major functions of the application, such as the Record Page, Progress Page, Reminder Page, Timelapse Page and Streak Page as well as another function tested is User Profile Page. Test objectives for each function include verifying accuracy and functionality, validating display and updates, ensuring correct navigation and displays, and testing various features such as reminder notifications, photo saving, and diet calculations based on user inputs. Then testing and results for each function were discussed, and the objectives of the project were evaluated with the successful completion of five objectives for a diet management application. The application features a diet management module that accurately calculates users' calorie intake per meal and updates daily consumption, a summary report system and a weight tracking tool to track users' weight on the Progress Page that monitors users' diet progress and displays estimated diet goal dates. Additionally, a reminder tool has been developed on the Reminder Page, allowing users to set and manage notifications for inputting food calorie intake, while the Timelapse Page enables users to save daily pictures and create a timelapse video showcasing their diet progress. Overall, all objectives have been fulfilled.

In conclusion, this chapter has provided a comprehensive overview of the system testing, performance metrics, and integration testing applied to the diet management application. The successful implementation and evaluation of various components have demonstrated the application's effectiveness in meeting its objectives. With a user-friendly interface and valuable features, the application offers a comprehensive solution to help users manage their diet, track their progress, and stay motivated throughout their weight loss journey.

CHAPTER 7 CONCLUSION AND RECOMMENDATION

7.1 Conclusion

To summarize, the project aims to develop a Diet Manager system into an Android mobile application. The project introduces some of the problems of the current existing diet app, which are methods of dieting not explained clearly to the user, users forgetting to input food intake into the app, and as well as users seeing the diet app as not interesting or fun. The motivation of the project is to implementation of better diet calorie counting features by providing detailed guides to educate users about dieting to solve the problems of the existing system, such as mentioned earlier. The proposed solution to the problems mentioned is to develop a diet manager system for individuals with weight concerns by implementing the system into an Android mobile application, including all dieting functions such as calculating the user's food calorie intake in every meal, a summary report system to monitor user dieting progress to keep track user diet goals, weight tracking tool to record user's weight and for the user to keep track of their weight, and a reminder tool for users to input user's food calorie intake when the users forget to. The project also contributes two novelties that can improve the diet manager app, which provides a tool for the user to create a timelapse of their diet progress by collecting and compiling all the pictures taken by the user daily throughout their dieting journey and providing an interesting streak system whereby user continue to redeem daily streak points and not to break the streak whenever the user does not over consume their daily calories limit. Both novelties aimed to increase users' continuous motivation to use the app and to have a happier user experience with the app.

7.2 Recommendation

First, the added food item can be improved because there's difficulty adding custom food items. One of the Diet Manager app's drawbacks is that it can make it difficult for users to add unique food items to their diet plans. This can be an issue for users who enjoy special or handmade dishes that are not readily available in the app's database. The app might include a function that lets users design their own food products and include them in their diet plan to get around this restriction. Users should be able to enter information about the meal, including its name, portion size, and calorie count. Additionally, the software can let users save their own food items for later use, saving them from having to be manually entered each time. Consider a situation where a user wishes to add a unique food item that isn't already in the app's database. The software must provide a function that enables users to manually enter meal names and nutritional details like calories, protein, carbohydrates, and fats. The user may then save this customized food item for later use and include it in their diet plan. For instance, the app may have a "Create Custom Food" option that directs users to a form where they can enter the specifics of their custom meal on the food search page.

Secondly, the Diet Manager calorie monitoring can be improved by including the user's daily physical activity that burns the user calories. Then the burnt calories can be included in the calorie counting calculation. The limited tracking of physical activity is another drawback of the Diet Manager app, which may not track this crucial component of leading a healthy lifestyle. The app can either include a feature that enables users to manually track their physical activity or interact with fitness trackers to overcome this issue. Users should be able to enter the kind of activity, how long it lasted, and how many calories they burnt using this function. The software can also offer advice and tips on how to balance one's food and exercise in order to reach fitness objectives. For customers who wish to manage their food and exercise in one location, the Diet Manager app can offer a complete solution by resolving these restrictions.

REFERENCES

- [1] M. Zečević, D. Mijatović, M. K. Koklič, V. Žabkar and P. Gidaković, "User Perspectives of Diet-Tracking Apps: Reviews Content," *JOURNAL OF MEDICAL INTERNET RESEARCH*, vol. 23, no. 4, p. 2, 2021.
- [2] R. Dotinga, "Do weight loss apps really work?," 18 November 2014. [Online]. Available: https://www.cbsnews.com/news/do-weight-loss-apps-really-work/.
 [Accessed 15 4 2022].
- [3] S. Zaidan and E. Roehrer, "Popular Mobile Phone Apps for Diet and Weight Loss: A Content Analysis," *JMIR mHealth and uHealth*, vol. 4, no. 3, p. 2, 2016.
- [4] "Windows install," Flutter, [Online]. Available: https://docs.flutter.dev/getstarted/install/windows.
- [5] "Java SE 11 Archive Downloads," Oracle, [Online]. Available: https://www.oracle.com/my/java/technologies/javase/jdk11-archivedownloads.html.
- [6] "Android Studio download archives," Android, [Online]. Available: https://developer.android.com/studio/archive.
- [7] "Samsung Android USB Driver," Samsung, [Online]. Available: https://developer.samsung.com/android-usb-driver.
- [8] H. M. S. A M Roza, "The harris benedict equation reevaluated: Resting energy requirements and the Body Cell Mass," *The American Journal of Clinical Nutrition*, vol. 40, no. 1, pp. 168-182, 1984.
- [9] "FoodData Central Download Data," United States Department of Agriculture,
 [Online]. Available: https://www.myfooddata.com/spreadsheets/MyFoodData-Nutrition-Facts-SpreadSheet-Release-1-4.xlsx.

(Project II)

Trimester, Year	:	Y3S3	Study week no. : 3
Student Name & ID	:	1902956	
Supervisor	:	Abdulkarim Kanaan Jebna	
Project Title	:	Diet Manager	

1. WORK DONE

[Please write the details of the work done in the last fortnight.]

Had transferred all contents from FYP1 report to FYP2 report. Completed some functions in application of the project.

2. WORK TO BE DONE

Continuing writing the FYP2 report. Work on the next Diet Manager's application functions.

3. PROBLEMS ENCOUNTERED

No major problems encountered.

4. SELF EVALUATION OF THE PROGRESS

- tors 110

Supervisor's signature

Unelik

Student's signature

(Project II)

Trimester, Year	:	Y3S3	Study week no. : 4
Student Name & ID):	1902956	
Supervisor	:	Abdulkarim Kanaan Jebna	
Project Title	:	Diet Manager	

1. WORK DONE

[Please write the details of the work done in the last fortnight.]

Completed some functions in application of the project.

2. WORK TO BE DONE

Continuing writing the FYP2 report. Work on the next Diet Manager's application functions.

3. PROBLEMS ENCOUNTERED

No major problems encountered.

4. SELF EVALUATION OF THE PROGRESS

Supervisor's signature

thetak

Student's signature

(Project II)

Trimester, Year	:	Y3S3	Study week no. : 5	
Student Name & ID):	1902956		
Supervisor	:	Abdulkarim Kanaan Jebna		
Project Title	:	Diet Manager		

1. WORK DONE

[Please write the details of the work done in the last fortnight.]

Completed half of the Diet Manager's application functionalities.

2. WORK TO BE DONE

Continuing writing the FYP2 report. Continuing the remaining half of the Diet Manager's application functions.

3. PROBLEMS ENCOUNTERED

No major problems encountered.

4. SELF EVALUATION OF THE PROGRESS

Supervisor's signature

thetak

Student's signature

(Project II)

Trimester, Year :		Y3S3	Study week no. : 7	
Student Name & ID	:	1902956		
Supervisor	:	Abdulkarim Ka	anaan Jebna	
Project Title	:	Diet Manager		

1. WORK DONE

[Please write the details of the work done in the last fortnight.]

Completed the remaining half of the Diet Manager's application functionalities. Completed half of the FYP2 report.

2. WORK TO BE DONE

Designing and creating the UI/UX design for the Diet Manager application.

3. PROBLEMS ENCOUNTERED

No major problems encountered.

4. SELF EVALUATION OF THE PROGRESS

Chat 115

Supervisor's signature

Unelish

Student's signature

(Project II)

Trimester, Year :		Y3S3	Study week no. : 9	
Student Name & ID	:	1902956		
Supervisor :		Abdulkarim K	anaan Jebna	
Project Title	:	Diet Manager		

1. WORK DONE

[Please write the details of the work done in the last fortnight.]

Completed half of the Diet Manager's application UI/UX design.

Completed the Streak functions of the Diet Manager.

Completed the UI/UX design for the whole Diet Manager application.

Tested all the functionalities of the Diet Manager to be working well with the implemented UI design.

2. WORK TO BE DONE

Making changes to the FYP report on the changed Streak functionality of Diet Manager.

Performing further testing on the Diet Manager application.

Continuing writing the FYP 2 report.

3. PROBLEMS ENCOUNTERED

No major problems encountered.

4. SELF EVALUATION OF THE PROGRESS

Everything is fine.

Supervisor's signature

Chelibh

Student's signature

Bachelor of Information Systems (Honours) Information Systems Engineering Faculty of Information and Communication Technology (Kampar Campus), UTAR

(Project II)

Trimester, Year :		Y3S3	Study week no. : 10	
Student Name & ID):	1902956		
Supervisor	:	Abdulkarim Ka	anaan Jebna	
Project Title	:	Diet Manager		

1. WORK DONE

[Please write the details of the work done in the last fortnight.]

All Diet Manager application's function has been completed. Completed testing on the application.

2. WORK TO BE DONE

Performing minor changes on the application to fix certain bugs. Completed the remaining half of the FYP report.

3. PROBLEMS ENCOUNTERED

No major problems encountered.

4. SELF EVALUATION OF THE PROGRESS

Supervisor's signature

thethat

Student's signature

(Project II)

Trimester, Year :		Y3S3	Study week no. : 11	
Student Name & ID	:	1902956		
Supervisor :		Abdulkarim K	anaan Jebna	
Project Title : Diet		Diet Manager		

1. WORK DONE

[Please write the details of the work done in the last fortnight.]

Completed and tested fully on all Diet Manager application's functionalities. Completed the FYP report.

2. WORK TO BE DONE

Compile and rearrange FYP report files according to format. And prepare project's application for submission.

Perform plagiarism check in Turnitin and have FYP2 report to be signed by Supervisor.

Preparing FYP report and project files to be submitted.

3. PROBLEMS ENCOUNTERED

No major problems encountered.

4. SELF EVALUATION OF THE PROGRESS

Everything is fine.

Supervisor's signature

Unelible

Student's signature

Bachelor of Information Systems (Honours) Information Systems Engineering Faculty of Information and Communication Technology (Kampar Campus), UTAR

(Project II)

Trimester, Year :		Y3S3	Study week no. : 12	
Student Name & ID):	1902956		
Supervisor	:	Abdulkarim Ka	anaan Jebna	
Project Title	:	Diet Manager		

1. WORK DONE

[Please write the details of the work done in the last fortnight.]

Completed FYP report and project's application.

Completed Plagiarism Check in Turnitin and have FYP2 report signed by Supervisor.

2. WORK TO BE DONE

Prepared report files, submit FYP report to portal.

3. PROBLEMS ENCOUNTERED

No major problems encountered.

4. SELF EVALUATION OF THE PROGRESS

Supervisor's signature

thetak

Student's signature

POSTER



Bachelor of Information Systems (Honours) Information Systems Engineering Faculty of Information and Communication Technology (Kampar Campus), UTAR

PLAGIARISM CHECK RESULT

ORIGIN	ALITY REPORT	
4 SIMIL	% 2% 1% 2% STUDENT	PAPERS
PRIMAR	Y SOURCES	
1	eprints.utar.edu.my Internet Source	1%
2	Submitted to Universiti Tunku Abdul Rahman Student Paper	<1%
3	thefastdiet.co.uk Internet Source	<1%
4	engineering.utm.my Internet Source	<1%
5	Ryan J. Faas. "iPhone for Work", Springer Science and Business Media LLC, 2010 Publication	<1%
6	www.jailbreakeye.com	<1%
7	Submitted to City University of Hong Kong	<1%
8	Submitted to Group Colleges Australia	<1%
9	Submitted to University of Hong Kong	<1 %

	10	Learn iOS App Development, 2013.	<1%
_	11	Submitted to Kingston University	<1%
_	12	Learn iOS 8 App Development, 2014.	<1%
-	13	uoil.blackstoneware.shop	<1%
-	14	Submitted to HELP UNIVERSITY Student Paper	<1%
-	15	fict.utar.edu.my Internet Source	<1%
	16	file.scirp.org	<1%
	17	Coetzee, Philna, and Dave Lubbe. "Improving the Efficiency and Effectiveness of Risk-Based Internal Audit Engagements : Improving the Efficiency and Effectiveness of Risk-Based Internal Audit Engagements", International Journal of Auditing, 2013. Publication	<1%
-	18	Hao Xu, Bin Xiao, Xiulong Liu, Li Wang, Shan Jiang, Weilian Xue, Jianrong Wang, Keqiu Li. "Empowering Authenticated and Efficient Queries for STK Transaction-based	<1%

	Computers, 2023 Publication	
19	Submitted to University of Technology, Sydney Student Paper	<1%
20	Submitted to Halesowen College	<1%
21	Sagar Sunkle, Suman Roychoudhury, Deepali Kholkar. "chapter 5 Authoring Models of Regulations", IGI Global, 2020 Publication	<1%
22	www.decodingdevops.com	<1%
23	Submitted to Asia Pacific University College of Technology and Innovation (UCTI) Student Paper	<1%
24	Submitted to CSU, San Jose State University	<1%
25	Submitted to University of Bedfordshire	<1%
26	Submitted to Wawasan Open University	<1%
27	forum.openframeworks.cc	<1%

28	eprints.kfupm.edu.sa	<1%
29	Beginning Scribus, 2015.	<1%
30	Rob Walters, Grant Fritchey. "Beginning SQL Server 2012 Administration", Springer Science and Business Media LLC, 2012 Publication	<1%
31	Wallace Jackson. "Android Apps for Absolute Beginners", Springer Science and Business Media LLC, 2017 Publication	<1%
32	www.berlmathges.de	<1%
33	Bruce Wade. "OS X App Development with CloudKit and Swift", Springer Nature, 2016 Publication	<1%
34	Jennifer Harder. "Chapter 2 Basic Warping Tools", Springer Science and Business Media LLC, 2023 Publication	<1%
35	Li Tung Tan, Raihanah Syamimi Mohammad Shahrizal, Muhammad Naqib'Aqil Mohd Nadzri, Norashikin Yahya, Suhaila Badarol Hisham. "Remote Patient Monitoring System", 2022 IEEE 5th International Symposium in	<1%

36	Rajesh k Sreekan Akhil Ra wheelch Confere Commu Publication	Kannan Meg ith, A. Govar j. "Wireless aair", 2017 4 nce on Adva nication Sys	galingam, Sarath rdhan, Chinta Rav gesture controlled th International anced Computing stems (ICACCS), 20	i Teja, d and)17	<1%
37	hopelive	e.hope.ac.uł	ĸ		<1%
38	laptop4	pro.vn			<1%
39	Wp.CO-a	ol.com			<1%
Exclud	le quotes le bibliography	Off On	Exclude matches	Off	

Universiti Tunku Abdul Rahman

Form Title : Supervisor's Comments on Originality Report Generated by Turnitin for Submission of Final Year Project Report (for Undergraduate Programmes)

Form Number: FM-IAD-005Rev No.: 0Effective Date: 01/10/2013Page No.: 1of 1

FACULTY OF INFORMATION AND COMMUNICATION TECHNOLOGY

Full Name(s) of Candidate(s)	LEE CHEE WAH
ID Number(s)	1902956
Programme / Course	Bachelor of Information Systems (Honours) Information Systems Engineering
Title of Final Year Project	Diet Manager

Similarity	Supervisor's Comments (Compulsory if parameters of originality exceeds the limits approved by UTAR)
Overall similarity index: <u>4</u> %	
Similarity by source	
Internet Sources: 2 %	
Publications: <u>1</u> %	
Student Papers: <u>2</u> %	
Number of individual sources listed of more than 3% similarity:0	
Parameters of originality required and line (i) Overall similarity index is 20% and	mits approved by UTAR are as Follows:

(i) Overall similarity index is 20% and below, and

(ii) Matching of individual sources listed must be less than 3% each, and

(iii) Matching texts in continuous block must not exceed 8 words

Note: Parameters (i) – (ii) shall exclude quotes, bibliography and text matches which are less than 8 words.

<u>Note</u> Supervisor/Candidate(s) is/are required to provide softcopy of full set of the originality report to Faculty/Institute

Based on the above results, I hereby declare that I am satisfied with the originality of the Final Year Project Report submitted by my student(s) as named above.

Signature of Supervisor

Signature of Co-Supervisor

Name: ____ Abdulkarim Kanaan Jebna

Name:

Date: _27/04/2023___

Date:

Bachelor of Information Systems (Honours) Information Systems Engineering Faculty of Information and Communication Technology (Kampar Campus), UTAR



UNIVERSITI TUNKU ABDUL RAHMAN

FACULTY OF INFORMATION & COMMUNICATION TECHNOLOGY (KAMPAR CAMPUS)

CHECKLIST FOR FYP2 THESIS SUBMISSION

Student Id	1902956
Student Name	LEE CHEE WAH
Supervisor Name	Abdulkarim Kanaan Jebna

TICK (√)	DOCUMENT ITEMS
	Your report must include all the items below. Put a tick on the left column after you have
1	checked your report with respect to the corresponding item.
N	Front Plastic Cover (for hardcopy)
	Title Page
	Signed Report Status Declaration Form
	Signed FYP Thesis Submission Form
\checkmark	Signed form of the Declaration of Originality
	Acknowledgement
\checkmark	Abstract
\checkmark	Table of Contents
\checkmark	List of Figures (if applicable)
\checkmark	List of Tables (if applicable)
	List of Symbols (if applicable)
	List of Abbreviations (if applicable)
	Chapters / Content
	Bibliography (or References)
\checkmark	All references in bibliography are cited in the thesis, especially in the
	chapter of literature review
\checkmark	Appendices (if applicable)
\checkmark	Weekly Log
\checkmark	Poster
\checkmark	Signed Turnitin Report (Plagiarism Check Result - Form Number: FM-IAD-
	005)
\checkmark	I agree 5 marks will be deducted due to incorrect format, declare wrongly
	the ticked of these items, and/or any dispute happening for these items in
	this report.
المناطح ماميا الممالة	(all a shirt) in the thesis (Dind to get her as the last mage)

*Include this form (checklist) in the thesis (Bind together as the last page)

I, the author, have checked and confirmed all the items listed in the table are included in my report.

thethak

(Signature of Student) Date: 28/4/2023